



# **Data-Driven Analytics Leveraging Artificial Intelligence in the Era of COVID-19: An Insightful Review of Recent Developments**

Abdul Majeed \* D and Seong Oun Hwang \* D

Department of Computer Engineering, Gachon University, Seongnam 13120, Korea \* Correspondence: ab09@gachon.ac.kr (A.M.); sohwang@gachon.ac.kr (S.O.H.); Tel.: +82-31-750-5327 (S.O.H.)

Abstract: This paper presents the role of artificial intelligence (AI) and other latest technologies that were employed to fight the recent pandemic (i.e., novel coronavirus disease-2019 (COVID-19)). These technologies assisted the early detection/diagnosis, trends analysis, intervention planning, healthcare burden forecasting, comorbidity analysis, and mitigation and control, to name a few. The key-enablers of these technologies was data that was obtained from heterogeneous sources (i.e., social networks (SN), internet of (medical) things (IoT/IoMT), cellular networks, transport usage, epidemiological investigations, and other digital/sensing platforms). To this end, we provide an insightful overview of the role of data-driven analytics leveraging AI in the era of COVID-19. Specifically, we discuss major services that AI can provide in the context of COVID-19 pandemic based on six grounds, (i) AI role in seven different epidemic containment strategies (a.k.a non-pharmaceutical interventions (NPIs)), (ii) AI role in data life cycle phases employed to control pandemic via digital solutions, (iii) AI role in performing analytics on heterogeneous types of data stemming from the COVID-19 pandemic, (iv) AI role in the healthcare sector in the context of COVID-19 pandemic, (v) general-purpose applications of AI in COVID-19 era, and (vi) AI role in drug design and repurposing (e.g., iteratively aligning protein spikes and applying three/four-fold symmetry to yield a low-resolution candidate template) against COVID-19. Further, we discuss the challenges involved in applying AI to the available data and privacy issues that can arise from personal data transitioning into cyberspace. We also provide a concise overview of other latest technologies that were increasingly applied to limit the spread of the ongoing pandemic. Finally, we discuss the avenues of future research in the respective area. This insightful review aims to highlight existing AI-based technological developments and future research dynamics in this area.

**Keywords:** artificial intelligence; COVID-19; data-driven analytics; privacy; epidemic; epidemiological investigations; epidemic containment strategies; healthcare; data lifecycle

## 1. Introduction

The recent pandemic of novel coronavirus disease 2019 (COVID-19) has changed our lives into a new normal where free-mobility, social gatherings at a large scale, and traveling seem impossible for the next couple of years. The COVID-19 has forced the closure of many cities and borders for a prolonged period of time. Furthermore, changes in the business/work hours and operating procedures of most organizations have completely changed. One of the biggest religious gatherings of the world at Mecca was cancelled or scaled down due to the pandemic last year [1]. Specifically, the whole world is going through an unanticipated and extraordinary challenge of COVID-19 [2]. Although there is a bright hope in terms of vaccine to end this pandemic, its distribution to underprivileged countries is a main challenge. Furthermore, rehabilitation of the healthcare system to pay ample attention to other existing diseases is also one of the main challenges in the near future [3]. In the absence of potential vaccine, one of the main technologies that played a critical role in combating the pandemic is artificial intelligence (AI) [4]. It can help in curbing the disease spread through contact tracing, social distancing, quarantine monitoring, trends analysis, symptoms reporting and analysis, symptoms clustering, symptoms severity



Citation: Majeed, A.; Hwang, S.O. Data-Driven Analytics Leveraging Artificial Intelligence in the Era of COVID-19: An Insightful Review of Recent Developments. *Symmetry* **2022**, *14*, 16. https://doi.org/ 10.3390/sym14010016

Academic Editor: Nikita Andriyanov

Received: 16 November 2021 Accepted: 18 December 2021 Published: 23 December 2021

**Publisher's Note:** MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



**Copyright:** © 2021 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). estimation, disease spread modeling, and alerting. We explain these services in detail in Section 3. A generic overview of application areas where AI has already demonstrated its effectiveness are shown in Figure 1. This study demonstrates AI applications in the context of the COVID-19 pandemic.



Figure 1. Overview of application areas of AI (Ref. [5]).

The impact and efficacy of AI techniques and models have been reported by many countries in curbing the disease spread. AI models have helped to identify the transmission routes of this disease and helped in mitigating the disease [6]. Further, the adoption of AI aided in recovering the economies from the low-levels with improved policies [7]. The adoption level of AI in each country was different. We present the latest finding about AI use with real-data until November, 2020 [8] and synthetic data from November, 2020 onward in the top ten countries across the globe in Figure 2.



Figure 2. Histogram of data about the use of AI techniques in ten countries (Ref. [8]).

From the results provided in Figure 2, it can be seen that the adoption of AI was higher in China, and this is the first country that curbed the pandemic spread quickly [9,10]. Besides the higher use of AI during this pandemic, the adoption of technical mechanisms (e.g., automated decision support systems, AI-driven diagnosis, and mobile doctors etc.) in the healthcare sector are relatively higher in China compared to other countries [11,12]. Therefore, AI-powered healthcare systems as well as other rigorous measures helped China to contain the spread of COVID-19 quickly. Furthermore, in those countries that adopted AI techniques at a smaller scale, the pandemic forced the closure of many facilities and activities [13]. Although AI played a vital role in this pandemic, many barriers were there in the adoption of AI, such as privacy and data manipulation, etc. In Figure 2, we chosen a sample of ten representative countries based on the origin of the pandemic, severity of disease, higher daily cases tally, digitization in the healthcare sector, and/or COVID-19

variants reported. In some countries (i.e., India, Pakistan, and Bangladesh), the covid cases were relatively higher, but adoption of the digital mechanisms in all these countries is significantly lower. That is why we did not include those countries in the analysis.

There are several studies that have covered this topic, especially AI role in the ongoing COVID-19 pandemic [14–18]. However, these studies have covered only general services of AI in COVID-19 context, and the main emphasis of most studies was on digital surveillance (or contact tracing). Furthermore, the data-driven analytics on actual data and AI essence from multiple perspectives have not been covered in prior studies. To cover these deficiencies, this study provides insightful coverage of the state-of-the-art studies that have devised ways to fight with COVID-19 pandemic leveraging AI. The main contributions of this article in the field of AI-based data-driven analytics in the COVID-era are summarized as follows.

- It covers the role of AI in COVID-19-era in six distinct regards such as epidemic containment strategies (ECS), epidemic data life cycle (EDLC), epidemic handling with heterogeneous sources data (EHHSD), healthcare-specific AI (HCSAI) services, general epidemic AI services (GEAIS), and drug design and repurposing (DDAR) against COVID-19 that have not been covered in the recent literature.
- It discusses the challenges involved in applying AI on the available epidemic data that is not in desirable form until present due to several problems (e.g., diverse formats, legislation, heterogeneous sources, and privacy concerns etc.).
- It elaborates the privacy issues that arise due to the person-specific data movement in cyberspace amid the ongoing pandemic.
- It provides a concise overview of the latest technologies other than AI that contributed in the fight against the recent pandemic through their innovative features.
- It discusses many state-of-the-art studies that have applied AI techniques in the ongoing COVID-19 pandemic for beneficence (i.e., greater good to save lives).
- It provides many state-of-the-art studies that have demonstrated the role of IoT based on heterogeneous data stemming from the ongoing pandemic to lower its effects.
- It discusses the synergy of AI with other emerging technologies in order to lower the effects of COVID-19 on the general public and economies.
- It provides the avenues of future research in the respective area keeping the latest technologies in loop.

The rest of this paper is organized as follows. Section 2 describes the prior research status and compare presented work results with related work. Section 3 provides the role of AI in fighting against COVID-19 through unique services in six different aspects. Section 4 discusses the challenges involved in applying AI on the COVID-19 data that is not in perfect until present. Section 5 summarizes the work, discusses emerging technologies role, AI synergy with other techniques, IoT-based developments in COVID-19 context, and provides promising future research directions. Finally, this article is concluded in Section 6.

## 2. Prior Research Status

In this section, we concisely present the contribution of previous studies, and compare proposed work results with related work. From the start of this deadly pandemic, AI has played a vital role in tackling it from a non-pharmaceutical interventions (NPI) point of view across the globe. The unique applications of the AI have paved the way to manage the resources well, and lowering the mortality rates through precise forecasting [19–21]. With the help of precise forecasting, extra care can be provided to the vulnerable people having underlying diseases, and treatment can be done on the regular basis [22,23]. Consequently, mortality rates and ICU admission can be prevented in most cases [24]. Arora et al. [25] discussed the potential applicability of AI in the development of early warning systems and accurate and timely forecasting about cases/mortality leveraging social media data. The study suggested that AI can be used in different aspects pertaining to COVID-19, but various issues like unavailability of the large datasets, ethical concerns, security and privacy, and computing resources remain challenging. Huang et al. [26] provided comprehensive

coverage of the AI in terms of clinical applications. For example, authors discussed the diagnosis of the COVID-19 via images, ultrasounds, chest scans, X-rays, lab indicators, electronic medical records, and lab indicators. Authors discussed AI role as experienced physicians for diagnosing COVID-19 robustly and accurately. We present the approach of Huang et al. [26] that was proposed in order to perform diagnosis of leveraging AI based on medical characteristics in Figure 3.



**Figure 3.** A flowchart of the AI methods employed for COVID-19 diagnosis and other relevant services: ML and DL were mainly applied in the medical characteristic to diagnosis the COVID-19 infection (Partially adapted from Huang et al. [26]).

Motta et al. [27] discussed the AI role from COVID-19 diagnosis and spread control point of views. The authors stressed the need of AI-powered decision systems to fight with the infectious diseases. Cave et al. [28] emphasized the need of using AI ethically while fighting with the COVID-19. Authors discussed four pillars of the biomedical ethics in order to get true benefits from the AI during crisis. The four pillars are (I) Beneficence, (II) Non-maleficence, (III) Autonomy, and (IV) justice. These pillars are an integral part of the healthcare settings in order to truly benefit from AI applications.

- Beneficence: It means that AI use should be beneficent (i.e., save lives) in managing the ongoing pandemic.
- Non-maleficence: It means the objective function of AI systems should be defined carefully in order to avoid unintended harms while managing the pandemic. For example, imposing strict self-isolation on elderly people may lead to mental issues.
- Autonomy: It means that people should be autonomous while controlling and endorsing the technologies including AI during the pandemic. For example, diagnostic support systems employed by healthcare workers in a pandemic should provide enough information about the uncertainty surrounding, and assumptions behind, a recommendation, so that it can be included into their professional judgment.
- Justice: It means that when AI systems are devised for a response to a COVID-19-like pandemic, difficult trade-offs between values could be incorporated. For example, decision about whether to employ centralized or decentralized app approach for data collection in order to manage pandemic should be based on justifiable grounds (e.g., involvement of diverse communities/stakeholders in decisions).

Leslie et al. [29] discussed the dark sides of the AI in terms of health inequity. Authors suggest that in order to reduce the inequalities, a collaboration between different stakeholders is paramount. An important perspective regarding the use of untested AI algorithms/methods in COVID-19 context is presented by the authors [30]. Authors suggested that in order to fully affirm the role of AI in saviour of the pandemic or future pandemics, we need to test the solutions with proofs. Chang et al. [31] concisely discussed the role of the AI from different perspectives including diagnosis to therapy. Authors discussed the role of the AI in three aspects such as epidemiology (predictions mainly), diagnosis, and therapy. According to the study findings, mismatch between epidemiology and data science need to be resolved in order to take advantage of the AI approaches for future endeavors. Vaishya et al. [32] analyzed the literature, and discussed the seven most significant AI applications (as shown in Figure 4) for the COVID-19 pandemic. Authors suggested that AI can play a dominant role in decision making and treatment consistency through robust algorithms. We fully agree with the contributions and significance of all studies cited above in the context of COVID-19 pandemic.



**Figure 4.** Overview of seven innovative AI applications in the context of COVID-19 (Adapted from Vaishya et al. [32]).

The advantages and significance of the proposed work compared to the prior studies are summarized as follows. (i) it provides insights about huge variety of data that is essential to fight with the COVID-19 leveraging AI from different perspectives, (ii) it discuss AI role in data life cycle and containment strategies that is not covered by any of the previous studies from broader perspective, (iii) it discusses substantial number of challenges comprehensively that hinder the applicability of AI methods in the ongoing pandemic, (iv) it provides the coverage of AI applications from multiple perspectives rather than one/two aspects, (v) it discusses the role of other emerging technologies with whom AI can be converged to serve the mankind in an effective way compared to the recent past, and (vi) it highlights actual AI applications based on the real-world data originating from the COVID-19 diagnosis or clinical practices.

## 3. Role of Data-Driven Analytics Leveraging AI in the Era of Covid-19

This section concisely presents the role of the data-driven analytics leveraging AI in the era of COVID-19. We categorize the coverage of AI applications/services in six regards such as epidemic containment strategies (ECS) that are in place as NPIs, epidemic data life cycle (EDLC) (aka data collection, storage, pre-processing, analysis, use, distribution, archival, and secure disposal) that is adopted in healthcare sector to fight with the infectious diseases, epidemic handling with heterogeneous sources data (EHHSD) as relying on a data from few sources is insufficient to fight with COVID-19, healthcare-specific AI services (HCSAIS) that can reduce the burden of healthcare workers, general epidemic AI services (GEAIS), and AI role in drug design and repurposing against COVID-19. These services are unique and emphasize the effectiveness of AI in COVID-19 era. Our aim is to highlight the significance of AI in COVID-19 context through relevant data and services. We identify data that relate with COVID-19 through analysis of the COVID-19 characteristics, and corresponding data. For example, SN tweets and comments that use the word COVID or related aspects such as quarantine, social distance, amid pandemic, and spread etc. are classified as COVID-19 data. In addition, some applications are specifically designed to collect and process data that relate with COVID-19. In some cases, data is collected in a proactive manner, and it can relate with COVID-19 when he/she tested positive. Furthermore, we mainly discuss

the additional data collected to fight the COVID-19 that varies from site to site as shown in Figure 9. Apart from these services in six regards, AI has been widely used in analyzing the vaccine distributions and other clinical aspects concerning COID-19. We describe each perspective and AI services in each perspective as follows.

#### 3.1. Perspective 1: Epidemic Containment Strategies and AI Role

From the beginning of the pandemic, each country of the world introduced certain strategies to contain the spread of COVID-19, including strict lockdown, cities and school closures, remote telehealth, closure of bars and clubs, and work from home, etc. Apart from these general containment strategies, many digital solutions based on the latest technologies for exposed people identification, close contact analysis, and compliance monitoring with the disease guidelines were also developed. We call such solutions epidemic containment strategies (ECS), and provide different AI-supported services in each ECS. We present the role of AI in seven different ECS that were extensively used in COVID-19 in Figure 5. Based on the extensive review of published studies, we found that AI remained a critical component of every ECS developed to contain the spread of COVID-19 [33–37]. Apart from the services cited in Figure 5, AI can play a vital role in alerting people stay away from the virus contaminated places pro-actively. In addition, it can be used to identify the focus group to whom COVID-19 can affect more due to underlying diseases. Hence, the role of AI in each ECS is vital and essential. Data reported in Figure 5 can be gathered from multitude of sources. For example, surveillance data can be employed for the contact tracing purposes [38]. We present an example of surveillance data based contact tracing example in Figure 6.

Epidemic Containment strategies	Major services provided by the AI in each epidemic containment strategy.
Contact tracing	[high risk people prediction, finding contaminated places/people, risk indexes]
Quarantine monitoring	[prediction/analysis of the quarantine violations, data analytics on regular basis]
Social distance monitoring	[analyzing the nature of contacts, social circle analysis, and flow modelling for better tracing]
Disclosing Patients/People data	[hotspot analysis, symptoms analysis, cluster analysis, lethality prediction, suspects ranking]
Symptoms and other data reporting	[data reduction, data pre-processing, data sparsity handling, and data skewness analysis]
Data collection at the time of checkups	[Minimal and need-based data collection, prediction about data disclosure risks/scores]
Data Analytics and Mining	[data compression and rules-based retrieval, and repositories storage, trends, forecasting]

Figure 5. Overview of AI-supported services in ECS in the context of COVID-19 pandemic.



**Figure 6.** Example of surveillance data based contact tracing for COVID-19 suspects finding. (1) Person A goes to work, bringing a Bluetooth-enabled cell phone with a digital key, which is used to communicate with other cell phones. (2) Person A comes in close contact with persons B, C, and D; all their cell phones exchange key codes with each other. (3) Person A later learns he is infected with COVID-19 and enters his updated status in the app. (4) By agreeing to share his recent status with the database, A instructs the app to send the data to the cloud service. (5) Meanwhile, B's, C's, and D's phones are regularly synchronising the cloud database to check the status of their users' close contacts. When B, C, and D discover that person A has reported himself infected, they all know they should get tested for the COVID-19 (Adapted from Hsu et al. [39]).

The mobile devices generated big data that can be used to monitor the people under quarantine [40]. In addition, GPS data can also be gathered for quarantine monitoring purposes [41]. In South Korea, health authorities usually call people on their cell-phone randomly, and such calls data can be used for the quarantine monitoring. The data to monitor social distance can be collected through Bluetooth technologies, video sequences, smart camera, and IoT platforms [42-45]. Data about the exposed/infected people can be gathered from the relevant clinics/diagnostic-centers. It can be shared with different entities to find the contacts of infected people [46]. COVID-19 symptoms and other related data can be gathered with the help of the wearable sensors, ambient tools, and smart phone technology [47,48]. Personal data at the time of the check-ups can be collected via automated/traditional forms. Subsequently, this data is used to rank the areas based on COVID-19 prevalence etc. Finally, CT and X-ray images data [49], ultrasound imaging data [50], text data [51], social media data [52], biomedical data [53], and big data [54] can be used during analytics. Apart from the data sources mentioned above for each ECS, data can be collected from heterogeneous sources for each category. For instance, analytics can be performed on data collected from variety of sources such as smart watches, sensors, mobile technology, CCTV cameras, GPS locations, Bluetooth devices, and smart write bands, to name a few. In Figure 5, the risk indexes are the quantitative values that denote the risk of being infected with COVID-19 based on gender, age, health status, city of residence, and knowledge of vaccine/treatment. This value is highly useful to evaluate infection risk accurately and taking preventive measures accordingly. AI can be used to rank the most influential indicators, and predicting the index accurately. The parameters used in each AI

model can be different. For example, if random forest is employed then parameters can be number of trees, variables required for tree split, sampling scheme, trees' complexity, and model type, etc. The choice of parameters and their values highly depends on the AI model chosen for the desired task.

## 3.2. Perspective 2: Epidemic Data Life Cycle and AI Role

In South Korea, if a person tests positive for COVID-19, then his/her contact details are collected to find the exposed people [55]. Different entities (i.e., police, mobile carriers, and credit card companies, etc.) collect contact data and process it in accordance with the specified procedures. This mechanism usually follows a lifecycle such as data collection, storage, pre-processing, analytics, use, archival, and deletion. We call this whole process an epidemic data life cycle (EDLC). We describe the role of AI in the EDLC in Figure 7.



Figure 7. Overview of AI-supported services in EDLC in the COVID-19 context.

AI can enable real-time decision making based on the collected data through EDLC. Since EDLC is essential to curb the spread of COVID-19, different AI mechanisms can be used at each phase of the EDLC. Although AI contributes significantly in all phases of the EDLC, geo-fencing of a certain area to contain the spread of COVID is one of the most useful applications. In geo-fencing, people of certain areas are put under home quarantine, and later they are monitored whether they are within the geo-fenced area or not using AI [56,57]. The example of a geo-fenced area (also referred as hotspot) with a dotted line is shown in Figure 8.



Figure 8. Example of the geo-fenced area.

In South Korea, health authorities usually keep track of the geo-fenced zones/areas and people living in respective areas through real-time location data processing via smart phone application. Furthermore, wrist bands were also used to monitor the quarantine violations in the geo-fenced areas. Data reported in Figure 7 can be collected from the infected individuals by either interviewing them or using the devices (i.e., cell phones) owned by them. Furthermore, every country has implemented various epidemic handling systems in the form of platforms, mobile apps, and integrated frameworks for supporting the EDLC. For example, South Korea has implemented an epidemic investigation support system (EISS) in which data from multiple companies (e.g., credit card, mobile carrier, and pharmacies etc.) is fed into it [58]. The EISS has an ability to collect and share the data with relevant and authorized entities. Similarly, Singapore Government asked their citizens to install a mobile app through which social interactions were recorded and processed with proper consents. Furthermore, data in EDLC can also be collected from the IoT devices and digital tools such as CCTV. The CCTV data have played a vital role in finding the COVID-19 suspects in South Korea. For example, when a sporadic cluster emerged at a gay club, then CCTV data has played a vital role to identify the people who may have been come into contact with infected people leveraging multiple CCTVs footage. We invite interested readers to gain more insight about type and nature of data processed in EDLC in previous studies [59,60]. In [59], authors discussed the consequences of COVID-19 on different people based on their working environments. In [60], authors discussed the importance of big data technologies in processing large scale data. Most of the AI models can handle the missing values present in a data. Furthermore, a simple and popular approach to address missing values related issues is data imputation. It employs statistical methods in order to estimate a value for a column from those values that are present, then replaces all missing values in the column with the calculated statistic. Furthermore, many AI models predict the missing values based on the original data statistics, and determine the missing values.

#### 3.3. Perspective 3: Epidemic Handling with Heterogeneous Sources Data and AI Role

Due to the nature of this pandemic, reliance on single data source, for example, relying solely on individual memory to figure out the contacts he/she has made in the past fourteen days or using popular SN data only to find vulnerable regions [25], etc.) has proven unsatisfactory in many countries across the globe. For example, in South Korea, manual epidemiological investigation (i.e., interviewing confirm patients about their travel information, facilities visits, and persons to whom they met etc.) has failed badly (in addition, it can slow down the containment of virus due to reliance on someone's memory and inaccuracies), and in addition to manual investigation, heterogeneous sources data collection and analysis helped to keep daily cases at a manageable level [61,62]. By using data acquired from different sources such as cellular network, credit card, surveillance cameras, and facilities-visits logs to find the exposed people is called epidemic handling

with heterogeneous sources data. Based on the extensive analysis of literature and technical developments [58,63], we classify the countries based on amount of heterogeneous sources data used to handle COVID-19 in four different categories in Figure 9. From Figure 9, it can be observed that South Korea used a huge amount and variety of data. Consequently, South Korea has better control on the ongoing pandemic without strict lock-down. In contrast, France has used less data, but the use of some apps was mandatory, therefore, heterogeneity in data was high. Middle eastern countries have primarily focused on monitoring people compliance rather than huge data collection. In Pakistan, the adoption of digital mechanisms in healthcare industry is relatively low, which is why only required data (e.g., voluntary reporting) was collected and processed during the ongoing pandemic. The heterogeneity in collected data in middle eastern countries and Pakistan are average and low, respectively.



Figure 9. Overview of heterogeneous data collected in different countries of the world.

The different data types listed in Figure 9 can be collected through combination of the digital and manual methods. For example, geolocation data can be collected using GPS sensors or Bluetooth devices. In South Korea, geolocation data of the confirmed patients was collected through mobile carriers. Symptoms and quarantine monitoring related data can be collected using low-cost sensors and/or calling people at random times and acquiring location in real time. Personal data can be collected through forms and websites, etc. Facility visit data can be obtained from logs maintained by each organization on daily basis, and historical diseases data can be obtained from hospital databases and mobilephone based surveys. Furthermore, travel data can be obtained from the travel agencies or airport staff. Recently, unmanned aerial vehicles have also been deployed to monitor people's compliance with the government guidelines. In addition, data is mostly collected prior to taking COVID test in South Korea. Furthermore, interviews and surveys are promising tools to acquire data. In some countries, cough sound, breathing patterns, blood samples, and temperature reading were taken through integrated platforms and sensors. Furthermore, to assess the spread, weather agencies and meteorologists also contributed to data collection. SN data have also paved the ways for generating symptoms taxonomy, and identification of new and related symptoms to COVID-19. Plenty of data collection methods have been comprehensively discussed by Hensen et al. [64]. We demonstrate the analytics to be performed on the heterogeneous data sources using AI in Figure 10. The analytics results can be extremely useful to keep the cases at a manageable level in order to lower the healthcare burden. Using data-driven analytics from different context can alleviate the pandemic's crisis.



Figure 10. Overview of AI-supported analytics on heterogeneous sources data.

Data reported in Figure 10 can be collected from a variety of devices, apps, and search engines, to name a few. IoT data can be collected from wearable sensors [65], contents, and other related data can be gathered from e-commerce websites [66], mobile data can be gathered from mobile carriers/service-providers, social media data can be collected from SN service providers [67], historical data can be gathered from the hospitals websites/repositories [68], medical images and sounds data can be collected with wearable devices or automated machines, and demographics data can be obtained from trusted clinics/hospitals [69]. The logistic regression based models can assist in identifying hotspots in any territory based on the certain environment parameters and underlying conditions [70].

## 3.4. Perspective 4: Healthcare-Related Services and AI Role

Besides AI use in data analytics, ECS, and EDLC described earlier, AI can be highly useful to assist in carrying out diagnosis and trend analysis to assist mankind in an effective way [71–73]. In this perspective, we discuss the possible use of AI in healthcare-related services that can have a direct impact on the virus virulence in any country. These services have unique utilities such as lowering hospital burdens [74], caring for elderly patients [75], separating the more risky people [76], and preventing people from being infected with COVID-19 [77]. To this end, we describe many healthcare-specific services of AI in Figure 11. We invite interested readers to the previous study for more details about each service in the context of ongoing pandemic [78].



Figure 11. Overview of AI-supported services in healthcare in COVID-19 context.

## 3.5. Perspective 5: General Epidemic Services and AI Role

In this perspective, we shed light on AI use in other sectors that were directly impacted by the COVID-19 pandemic. For example, predictions about when aviation industries can return to normal [79], how much transportation use reduced in each country due to COVID-19 [80], and recommendations of show/music to alleviate people's stress during this pandemic [81]. AI can play a vital role to access the risk and challenges of any sector during these unprecedented and unanticipated times [82]. In some sense, AI is lowering the human involvement in many sectors through automated and real-time decision making abilities [83]. For example, in south Korea, AI-powered robots were installed at the airports that were carrying similar tasks (i.e., temperature checking, mask status analysis and alerting, social distance monitoring and alerting in case of breaches, and preventing cluster formation of people at one place etc) as humans do [84]. In the new normal, AI use can possibly increase in many diverse sectors. For example, performing analytics using AI based on spatial-temporal data can be handy to predict and prevent future pandemics [85]. In addition, AI experiences can be applied to other epidemics to fight them effectively. We present general epidemic related services of AI in Figure 12.

AI uses in all six perspectives lay a solid foundation for future studies in the same area. It enables researcher and developers to understand the applicability of AI in different contexts. Furthermore, it provides conceptual foundations of AI use in different aspects related to pandemic and AI role in each aspect. Furthermore, these concepts can be exploited to devise new techniques for each services. In addition, there is a chance of improvising each AI technique on COVID-19 data that is relatively new and requires ample work to make sense of it.





in buildings

Figure 12. Overview of AI-supported general services in COVID-19 context.

## 3.6. Perspective 6: AI Role in Drug Design and Repurposing against COVID-19

Besides the AI use in the five different perspectives cited above, AI/ML has also been extensively used in computer-aided drug design and repurposing existing drugs against COVID-19 receptor proteins [86,87]. Monteleone et al. [88] discussed the role of AI in drug repurposing with therapeutics analysis for treating infected individuals with COVID-19. In this regard, we summarize the findings of recent SOTA studies in Table 1.

Table 1. Summary of the AI uses/applications in designing drugs and repurposing existing drugs against COVID-19.

<b>D</b> _(	Discussions about AI Use in the Era of COVID-19 along with Models Details and Purpose Achieved in the Context of COVID-19.			
Kei.	AI Technique Used	Purpose in the Context of Designing Drugs and Repurposing Existing Drugs		
Zhou et al. [89]	Fully connected feedforward neural network (FNN)	Drug repurposing for precision medicine and personalised treatment		
Walters et al. [90]	Quantitative structure-activity relationships (QSARs)	Drug discovery by predicting the physical properties and biological activity of molecules		
Patronov et al. [91]	Deep neural networks (DNN)	AI-based generative models for drug design to combat the COVID-19		
Arora et al. [92]	Deep neural networks (DNN)	Protein synthesis, molecular changes, time management in laboratory for drug dis- covery		
Bhati et al. [93]	ML integrated with PB	Sampling of relevant chemical space for target proteins analysis to make pandemic drugs		
Kabra et al. [94]	Combined AI approaches	Finding possible drug candidate to treat COVID-19 patients with antiviral drug		
Bai et al. [95]	Genetic algorithm	3D drug design of protein targets for treating COVID-19 patients		
Liu et al. [96]	Graph convolutional network (GCN)	Drug repositioning framework to quickly identity potential drugs for COVID-19		
Delijewski et al. [97]	Gradient boosting tree (GBT)	Identification of zafirlukast as one of the repurposing candidates for COVID-19		
Haneczok et al. [98]	Graph-CNN	Prediction of molecular property and identification of SARS-CoV-2 3CLpro inhibitors		

# 4. Data Challenges While Applying Ai in the Era of Covid-19

AI has played a vital role in many aspects to curb this disease spread. Meanwhile, the true applications were hindered by data that is not perfect and complete in many regards. We present a taxonomy of the challenges related to data that can possibly hinder AI use in Figure 13. These challenges need to be resolved to truly benefit from AI techniques. Some of these challenges can be solved by AI itself. For example, multi-model mechanisms can be used to make sense of heterogeneous sources data, feature engineering can be employed to filter redundant/less-important data before applying AI, data manipulation can be reduced by using advanced form of AI (i.e., federated learning), synthetic data can be generated by projecting original data using AI, and dimensionality can be reduced using many AI techniques.



Figure 13. Challenges involved in applying AI in the COVID-19 era due to data issues.

Concise description about each challenge, indicators, and possible solutions are described as follows.

Heterogeneity of data styles: During this pandemic, the data of diverse types is being collected. For example, in South Korea, when a person is confirmed to have a COVID-19, his/her data (contacts, place, demographics, and 14 days visits to every place, etc.) is collected in heterogeneous formats. For example, the routes information can be in graph form, buying items can be in tabular form, and facilities he/she has visited can be in matrix form. Hence, fusing this heterogeneous data from different contexts to find the potentially exposed people is very challenging. It requires parsers and unified format conversion to deal with diverse data that is very challenging.

- Heterogeneous sources data handling: During this pandemic, the data can originate from different sources. For example, in South Korea, fine-grained and sufficiently detailed data is collected to curb the disease spread. For example, healthcare sectors are constantly acquiring data from law and enforcement agencies, credit card companies, and SN etc. Hence, handling this heterogeneous sources data during pandemic time is very challenging. It requires interfacing and transparent policies and data-driven approaches to address this challenge.
- Data manipulation and misuse: During this pandemic, a huge amount of personal data is being collected on a daily basis. For example, mobility data, trajectory information, buying data, and social interactions, to name a few. Hence, it increases the chance of manipulation and misuse. It requires legal, organizational, and technical measures to address this challenge.
- Data volume: During this pandemic, a huge amount of data is being collected on a daily basis about people. For example, in South Korea, before entering any facility, data is being collected along with explicit identity information. Similarly, cellular networks data is used to perform crowd analysis, and identifying people at controversial places. Hence, handling of such higher volume of data is challenging, and it requires usage of high performance computing model. Additionally, it requires low-cost reduction and compression techniques to address this challenge.
- *Lack of data knowledge*: During this pandemic, a huge amount and diverse types of data is being collected from heterogeneous mediums. AI experts and analytics companies have limited knowledge of data structures and formats. For example, temporal and spatial data can be in different formats and styles. Hence, converting such a data into consistent styles prior to AI application is very challenging. It requires domain expertise and visualizations techniques to address this challenge emerging in pandemic times.
- Data convergence issues: In this pandemic, data was being collected from different sources, and in different styles (e.g., graphs, matrix, tables, etc.). Correlating/converging different subjects data gathered from different sources and styles is challenging. For instance, analyzing the characteristics of each subject based on data he/she produces or consumes using different sources is an extremely difficult task. It requires preprocessing and similarities-based approaches to address this challenge emerging in pandemic times.
- *Inadequacy of metrics*: In this pandemic, the majority of data analytics was performed using existing metrics that yields imprecise results considering the huge dynamics of COVID-19. For instance, analyzing disease spread based on daily cases and ambient conditions is difficult in the absence of desired metrics. It requires new metrics or amendments in the existing metrics to make them more suitable for use.
- Lack of truthful data: In this pandemic, huge variations were observed in each territory regarding the disease severity, symptoms combination, and virus effect on the different age groups. For example, disease characteristics observed in South Korea exhibit large differences to those observed in Japan. Hence, to clearly understand the disease dynamics, there is a lack of truthful data, although some companies/researchers generated synthetic data that is close to the original data for understanding/modeling of the COVID-19 dynamics. Moreover, synthetic data may yield imprecise results in the absence of evidence-based truthful data [99]. This challenge can possibly be solved through data sharing with domestic and international firms, and analyzing data with advanced AI techniques.
- Mishandling in finding exposure of contacts: To accurately find the contacts of an infected
  person, close monitoring of all subjects in outdoor environments is paramount. For
  example, it requires monitoring of who met with whom? for how long he/she met?
  what was the nature of contact (e.g., had dinner/lunch or just crossed), whether he/she
  was wearing masks perfectly or not? and how often he/she met with each other. To
  capture and analyze all these aspects with fine-grained data collection about each
  subject can be highly difficult, and it can lead to hidden/silent transmission of COVID-

19. This challenge can possibly be solved through data collection from heterogeneous sources, and analyzing data with advanced AI techniques and integrated platforms.

• Data collection in a fine-grained manner: Due to the nature of this pandemic (i.e., spread through close contact), data about people should be collected in a fine-grained manner. Meanwhile, in some countries, data protection laws are in place, therefore, fine-grained data collection is not allowed. Due to which virus can spread, containment is not easy at all. For instance, the recent pandemic spread at a wider scale in European nations due to general data protection regulation (GDPR), which restricts data collection about subjects without their explicit permission. This challenge can possibly be solved through amendments in laws considering the severity of the virus for public safety.

## 5. Discussion on AI and Latest Technologies Use and Future Research Directions

In this section, we concisely discuss AI use/services in COVID-19 context based on actual data and purpose of using AI, latest technologies that have been used in the era of COVID-19, synergy of AI with other emerging technologies, and promising research directions for future endeavours.

## 5.1. Discussion on AI Use in the Context of COVID-19

So far, we have reported the coverage of AI-based analytics in the COVID-19 era from six perspectives such as ECS, EDLC, EHHSD, HCSAIS, GEAIS, and AI role in drug design and repurposing against COVID-19. We have rigorously and thoroughly analyzed the scope of AI in all six aspects. This concise overview will enable future research in this area with clear directions/gaps. Specifically, we highlighted the data related challenges that need to be resolved to yield higher adoption of AI, as AI has already demonstrated effectiveness in many aspects related to the COVID-19 [100–105]. Thus, we summarize AI use based on actual data used/processed, AI models applied, and purpose/service achieved in COVID-19 context in Figure 14.



Figure 14. Practical uses of AI to fight with the ongoing pandemic.

Apart from AI unique services cited above, they can be extremely useful in preventing diseases by predicting high risk facilities, and identification of lethal combination based on

geographic data that can lead to a higher number of deaths. Furthermore, AI can be handy to analyze protein sequences that can assist in developing potential vaccines for future pandemics. Furthermore, many unique aspects such as robustness, efficiency, large data handling, and reduction in time and cost make AI more attractive for many applications in the healthcare sector. In many countries, AI has been integrated as a main module with decision support systems (DSS) to efficiently fight this pandemic. Apart from the AI services explained above, we describe various promising applications of the AI in the era of COVID-19, keeping AI techniques used in loop in Table 2.

**Table 2.** Comprehensive overview of the AI use/applications in the era of COVID-19 discussed in recent SOTA studies.

P-(	ng with Models Details and Goals Achieved in the Context of COVID-19.	
Kef.	AI Technique Used	Purpose in the Context of COVID-19 Pandemic
Pinter et al. [106]	Multi-layered perceptron	Predictions of mortality rate and infected cases
Aminu et al. [107]	Deep neural networks	Detection of people with COVID-19
Nagar et al. [108]	Ensemble techniques Extreme Cradient Beesting (VCBeest)	Forecasting of patient survival probability
Ashraf et al [110]	Machine & deen learning models	Predict the severity of disease or chances of death
Shah et al [111]	Convolutional neural network (CNN)	COVID-19 detection from X-ray images
Prakash et al. [112]	Autoregressive Integrated Moving Average	Impact analysis of various policies
Rathod et al. [113]	AI Prediction models	Effective crisis preparedness and management
Ullah et al. [114]	Logistic Regression and Support Vector Machine	Classification of patients with/without COVID-19
Rathod et al. [115]	SVM, RProp, and Decision tree	Detection of abnormal data for effective analysis
Hu et al. [116]	Spectral Clustering (SC) algorithm	Feasible analysis model for the treatment & diagnosis
Rashed et al. [117]	Long short-term memory (LSTM) network	Provides public awareness about the risks of COVID-19
Singh et al. [118]	ResNet152V2 and VGG16 CNN	Reduce the high false-negative results of the RI-PCR
Saverino et al. [119]	Convolutional Neural Network (CNN)	Detection of COVID 10 second in multicarlages
Malla of al [121]	Ensemble deen learning model	Real-time continent analysis of COVID-19 data
Lella et al [122]	Convolutional Neural Network (CNN) model	Respiratory sound classification for patient identification
Haleem et al. [123]	Artificial neuronal networks (ANN)	Predictions of survival of COVID-19 patients
Hashimi et al. [124]	Deep learning models	Tracking and identifying potential virus spreaders
Amaral et al. [125]	Artificial neuronal networks (ANN)	forecasting and monitoring the progress of Covid-19
Zgheib et al. [126]	Collection of ensemble learning methods	Detecting COVID-19 virus based on patient's demographics
Ferrari et al. [127]	Bayesian framework	Predictions about the behavior of the COVID-19 epidemic
Almalki et al. [128]	COVID Inception-ResNet model (CoVIRNet)	Automatic diagnosis of the COVID-19 patients
Umair et al. [129]	VGG16, DenseNet-121, ResNet-50, and MobileNet	diagnosis of the virus at early stages via X-rays and transfer learning
lamagusko et al. [130]	EpiEstim framework	Analysis of the population's mobility during the COVID-19 pandemic
Hussain of al [132]	Ensemble learning methods (RF, SVM, and ANN)	Short-term and accurate prediction of effective reproduction number ( $K_f$ ) Analysis of public attitudes on Twitter & Eacebook toward COVID-19 vaccines
Kumari et al. [132]	Combination of multi class SVM and CNN models	Contact less authentication system and face mask identification
Talahua et al. [134]	OpenCy's face detector and MobileNetV2 architecture	identifying whether people are wearing face masks or not
Yu et al. [135]	GCNN ResGNet-C under ResGNet framework	Effective diagnosis of COVID-19 from lung CT images
Nayak et al. [136]	Lightweight and robust CNN scheme	Faster and accurate diagnostics of COVID-19 patients
Bekhet et al. [137]	Lightweight CNN architecture	Recognizing COVID-19 patients with a 96% accuracy
Keicher et al. [138]	Lightweight clustering method	Patients outcomes prediction admission to ICU, need for ventilation and mor- tality
Alshazly et al. [139]	Deep network architectures and transfer learning strategy	CT images-based diagnosis of COVID-19 infected people in an automated way
Carvalho et al. [140]	Convolutional features & genetic algorithms	Screening and diagnosis of COVID-19 patients
Fu et al. [141]	Lightweight DenseANet architecture	Distinction between pneumonia and COVID-19 patients using CT images
Bougourzi et al. [142]	Pre-trained XG-boost classifier	Analysis of sensitivity of the COVID-19 patients from CT images data
Song et al. [143]	Details relation extraction neural network (DRENet)	Person-level diagnoses of COVID-19 using CT images
Alruwalli et al. [144]	Inception-KesiNetV2 deep learning model	Visualization of the lungs' infected regions using CAR images
Wang et al. [145]	Inception transfer-learning model	19
Jha et al. [146] Abbas et al. [147]	logistic regression, SVM, Random Forest, and QSAR DeTraC deep convolutional neural network architecture	Robust drugs discovery and extraction of features combating COVID-19 Classification of COVID-19 chest X-ray images
Sedik et al. [148]	CNN & convolutional long short-term memory (ConvL-STM)	AI-powered COVID-19 detection system using X-ray and CT data
Bhardwaj et al. [149]	Inceptionv3, DenseNet121, Xception, and InceptionRes- Netv2	Quick and highly accurate automated COVID-19 detection
Muneer et al. [150]	Netv2 Deep hybrid NN models (GCN-GRU and GCN-CNN)	Prediction of RNA degradation from RNA sequences
All et al. $[151]$	Histogram-oriented gradient (HOC) and CNN	Detect of COVID-19 from the chect Y-ray images using model fusion
Raii et al [153]	Convolution Neural Networks using medical modalities	Robust detection of the virus by using the pre-trained models
Teli et al [154]	shallow and simple CNN-based approach, named TeliNet	Robust classification of CT-scan images of COVID-19 patients
[acobs et al. [155]	Generative deep learning models	Small molecule drug design using scalable deep learning for COVID-19
Madhavan et al. [156]	Res-CovNet: A hybrid methodology	Classification of multiple diseases using X-ray images
Shorfuz et al. [157]	IoT-enabled deep learning-based stacking model	Analysis of chest CT scans for diagnosis of COVID-19 encounters
Shankar et al. [158]	Cascaded recurrent neural network (CRNN) model	Detection and classification of the existence of COVID-19
Saranya et al. [159]	Recurrent NN utilized the TensorFlow Keras framework	COVID-19 mortality prediction using electronic health records
Alhudhait et al. [160]	CNN model built on DenseNet-201 architecture	Determination of COVID-19 pneumonia from X-ray images
Aboutalebi et al. [161]	COVID-Net CXK-S, a convolutional neural network	Predicting the airspace severity of a COVID-19 positive patients
∠nao et al. [162]	convolutional neural network (CNN)	COVID-19 Identification from a small subset of training data

## 5.2. Discussion on Latest Technologies Used to Fight with COVID-19 Pandemic

Beside AI, numerous other latest technologies such as blockchain (BC), federated learning (FL), few short learning (FSL), robotics, and confidential computing (CC) have also played a vital role in this pandemic. For example, BC technology has promising characteristics such as transparency, immutability, verifiability, and privacy-preservation which makes it suitable for data sharing securely between different entities [163]. In addition, it can be used to monitor the personal data flows in the healthcare environments. FL is an emerging technology in which data in not shared, instead model results are shared only. It can be highly beneficial to ensure individual privacy. Furthermore, it enables model results sharing at a wider scale without privacy disclosures [164]. FSL enable machine learning model's training from a few data samples. It has a lot of applications in analyzing the dynamics of COVID-19 by extracting knowledge using a limited data [165]. Robotics were used to deliver the test samples from testing sites to hospitals, and many other innovative applications [166]. In addition, they were also used to check people's temperature in the streets. Some countries used robotics to monitor people's mobility during the rush hours. CC techniques were employed to secure the personal data since it can use the data without accessing actual values. All these technologies have played a vital role in serving mankind during this deadly pandemic. We summarize the role of the seven latest technologies that were used in this pandemic as follows.

- Blockchain (BC): The BC technology has been widely used in addressing the challenges of privacy in healthcare sectors [167,168]. The unique capabilities of the BC such as decentralization, transparency, immutability, and traceability makes it useful for alleviating privacy problems of data storing, distribution, and utilization phases. BC has been rigorously used in this pandemic for transparency and verifiability related purposes [169].
- Decision Support Systems (DSS): DSS can play a vital role in lowering the burden of the medical staff [170]. They can be extremely useful for the ETL (extract, transform, and load) purposes and performing the desired screening tasks in an automated ways. DSS can be very helpful in lowering the burden of healthcare workers and planning resources accordingly.
- Explainable AI: It is very recent technology with a wide range of practical applications in the diagnosis and analytics [171]. It can be extremely useful in identifying the hidden routes of disease transmission and vulnerable communities analysis.
- Internet of things (IoT): IoT has revolutionized the medical sector with unique abilities
  of remote monitoring, connected healthcare, and constant tracking [172]. IoT can
  be highly useful in symptoms reporting, remote analysis of patients, and patients
  monitoring in ICUs, etc.
- Confidential computing and zero knowledge proofs: Both these techniques have higher utility in data distribution with different stakeholders [173]. These techniques enable data utilization with higher privacy guarantees. These solutions can be widely acceptable to address the privacy implications of the data distribution with domestic and international researchers.
- Natural language processing: It can be highly useful in the analytics phase of the EHS for symptoms extraction and sentiment analysis [174]. It can also be useful for symptoms clustering and forming a unified taxonomies of epidemic diseases.
- Search Engines (SE): The SE has played a vital role in devising the common symptoms
  of the infectious diseases, and it can assist in identifying the origin of pandemics. The
  tools can be employed to recommend helpful tips to the people to reduce the chaos
  created by pandemic.

In order to fight COVID-19, every country has implemented digital solutions and launched many projects. For instance, South Korea has implemented an integrated platform named epidemic investigation support system (EISS), in which data about infected patients is collected and shared with relevant agencies [61]. Furthermore, South Korea has implemented many smartphone apps for contact tracing, quarantine monitoring, and logging with the help of different ministries [175]. China has used plenty of AI techniques and DSS in response to COVID-19 [176]. Pakistan has launched a web-based portal to report the statistics of virus, and multi-criteria based data-driven testing strategy [177]. Singapore has implemented an app for social interactions recording that can be used later if someone get infected with COVID-19 [178]. Some countries have used AI and drones etc. to curb the spread of COVID-19. In the U.S., national COVID cohort collaborative (N3C) project is being launched in which multiple organizations are collaborating on clinical data related to COVID-19. In addition, the N3C project aims to answer important research questions that in return will assist to combat the COVID-19 pandemic [179]. Many countries have developed a variety of digital solutions during this pandemic to keep their citizens safe [180–183].

IoT and various smart sensing technologies are also playing a key role in the pandemic arena, leveraging COVID-19 pandemic related data for various purposes such as remote monitoring, quarantine management, and symptoms reporting [184–186]. A number of survey studies have discussed the IOT role in the ongoing pandemic [187,188]. In this regard, we summarize the role of IoT in COVID-19 pandemic reported by the SOTA and recent studies in Table 3.

**Table 3.** Summary of IoT and various smart sensing technologies role in fight against the COVID-19 pandemic.

<b>D</b> _(	Discussions about AI Use in the Era of COVID-19 along with Models Details and Purpose Achieved in the Context of COVID-19.		
Kei.	Data Sources	Purpose Achieved in the Context of Lowering the Effects of COVID-19 on General Public	
Sharma et al. [189]	Wearable sensors	Timely and accurately predicting COVID-19 positive cases to control the spread of COVID-19 pandemic	
Khan et al. [190]	Camera sensors	Monitoring and countering the spread of ongoing pandemic using Io1 sensors data in close indoor spaces	
Awotunde et al. [191]	Sensing technologies	Advising patients about their health conditions preventive measures suggestions to saving lives amid the pan- demic	
Abdulkareem et al. [192	] Medical devices	AI and IoT based clinical decision support systems for COVID-19 pandemic handling in smart hospitals	
Jayachitra et al. [193]	Handheld devices	IoT-based cognitive system with 100% prediction accuracy of COVID-19 infection using multimodal data	
Herath et al. [194]	Thermal cameras	IoT-based system to detect & control the ongoing pandemic inside the hospital environment	
Mukherjee et al. [195]	Medical devices	IoT-cloud-based healthcare predictive model in order to quickly detect COVID-19 using eKNN	
Akbarzadeh et al. [196]	Wearable sensors	Notifying end-users when breaking the social distance guidelines in the situation of COVID-19 pandemic	
Petrović et al. [197]	Sound sensors	A cost effective IoT-based practical solution for reducing the spread of COVID19 in indoor settings using cough sounds	
Alamri et al. [198]	IoT sensors	Providing real-time information to the users about potential events that can affect the public transport in COVID- 19 times	
Poongodi et al. [199]	COVID-specific sensors	A robust health-based fully connected IoT systems in order to strengthen full COVID-19 administration using location data	
Kent et al. [200]	Mobile sensors	IoT-based solution for hospitals in order to improve health monitoring and providing timelier healthcare for pa- tients	
Krishnan et al. [201]	Multiple sensors	Checking the availability of the mask in initial stage and monitoring the students' temperature in the latter stage	
Mylonas et al. [202]	IoT sensors	Analyzed the effects of COVID-19 pandemic on a multiple schools in Greece for monitoring energy and noises	
Bhowmick et al. [203]	IoT sensors	Process and help us monitor the health of older people in clouds based on different medical IoT sensors data	
Herath et al. [204]	IoT sensors	Prevention of the COVID-19 pandemic using IoT-based platform in a smart city environments	
Herath et al. [205]	IoT sensors	Monitoring the symptoms of COVID-19 infected patients, and detecting the patient's activities using mobile app	
Lastovicka et al. [206]	IR and ultrasound	Contactless solution for automatic induction of disinfection intelligent hand sanitizer to lower spread of COVID-19	
Rajasekar et al. [207]	RFID tags	IoT-based automated tracking and tracing method for identification of the possible contacts of COVID-19 patients	
Alhmiedat et al. [208]	Wearable sensors	Slowing the spread of COVID-19 locally and across the country by allowing individuals to maintain social dis- tances with others	

#### 5.3. Synergy of AI with other Emerging Technologies in the Context of COVID-19

In recent years, AI has been increasingly used in combination with emerging technologies such as IoT, IoMT, cloud computing, fog/edge computing, and federated analytics, to name a few for accomplishing multiple goals [209,210]. Ahuja et al. [211] discussed AI use from three different perspectives such as drug discovery, public communication, and integrative medicine in the context of COVID-19. Márquez et al. [212] discussed the joint use of AI and big data in the era of COVID-19. According to the authors, these synergies between disruptive technologies can facilitate obtaining relevant data that, in return, is helpful for health-related decision-making. Anjum et al. [213] discussed the emerging technologies to fight the COVID-19 pandemic. The authors have discussed the role of AI and IoT-assisted drone technology, these synergies between emerging technologies can pave the way to fight future infectious diseases using technology. Ahmad et al. [214] discussed the role of AI in COVID-19 pandemic by performing analytics on data stemming from COVID-19 pandemic. The authors discussed the synergies between various emerging technologies for healthcare decision support leveraging IoT sensors. Lainjo et al. [215] discussed the big data and AI synergies that have helped various nations in improving the pandemic situations and reducing the adverse impacts of COVID-19 on economies. Bazel et al. [216] discussed the synergies between AI and three other emerging technologies (i.e., IoT, blockchain, and big data technologies) in healthcare in order to prevent the spread of COVID-19. Swayamsiddha et al. [217] presented a comprehensive analysis of AI-aided detection of COVID-19 using heterogeneous sources of data (i.e., AI-enabled imaging). Deshpande et al. [218] discussed the promising applications of AI leveraging audio signals data. The authors have demonstrated the diagnosis and screening of COVID-19 patients using audio-based analysis. Despite these promising applications, AI is a promising solution for supply chain management in the post COVID-19 era [219]. A comprehensive discussion on synergy between blockchain and AI, along with their benefits and limitations, has been reported in the recent literature [220]. Recently, Deepti et al. [221] demonstrated the gamut of synergistic applications including AI, cloud-enabled IoT, connected sensors and actuators, and ubiquitous Internet to form connected communities that, in return, can help to fight the COVID-19 pandemic [221]. Furthermore, AI is an important pillar of industry 5.0 [222,223]. A relatively new concept, artificial intelligence of things (AIoT), has emerged as a new concept for addressing the potential limitations of IoT in healthcare 4.0 [224]. Considering the promising applications of AI in the healthcare sector, its synergy with other technologies is likely to increase in the near future for accomplishing multiple goals. Hence, it has become an emerging avenue of research in recent years to serve the mankind effectively.

For the convenience of readers, we provide the summary of the important AI-related developments (adapted from [225]) reported in the published surveys in Figure 15. The analysis presented in Figure 15 shows that AI has contributed significantly to multiple areas (e.g., drug design and development, diagnosis, and surveillance, to name a few). On the other hand, this analysis can enable researchers to contribute more to areas that have been given less attention in previous studies.



Representative AI application reported in the published surveys

Figure 15. Statistics of AI related developments reported in the prior surveys [225].

Besides the unique AI applications cited above, many surveys on AI topics have been published, and each survey has tried to demonstrate the AI uses from different perspectives. To aid subsequent research in this regard, we systematically summarize the findings of most recent surveys that have focused on AI applications in the context of COVID-19 in Table 4. The analysis presented in Table 4 clearly demonstrates the existing developments in review articles. Our review is an enhanced version of these surveys as we discuss AI's role more broadly compared to these studies. This analysis is helpful to quickly and conveniently grasp the research status of AI in the COVID-19 era.

**Table 4.** Key findings of most recent surveys that have focused on AI applications in the context of COVID-19.

Ref.	Pub. Year	Review Type	Key Findings Concerning COVID-19 Pandemic
Nadeem et al. [226] Abd-Alrazaq et al. [227] Raza et al. [228] Chen et al. [229] Enughwure et al. [230] Chiroma et al. [231] Bullock et al. [232]	2020 2020 2020 2020 2020 2020 2020 202	Literature survey Scoping review Meta-analysis Rapid review Systematic review Bibliometric study Literature review	Discussion of AI applications and data sources to fight against Covid-19 via technology Discussion about AI technology use during the ongoing COVID-19 pandemic Discussion from broad spectrum of AI to combat COVID-19 by analyzing current SOTA studies Discussion and review of the critical aspect of AI applications for COVID-19 era Analyzed 15 SOTA studies and showed AI has many potentials in combating COVID-19 pandemic Discussion on ML-based technologies to fight the COVID-19 pandemic from multiple perspectives Reviewed many datasets, resources, and tools required to facilitate AI research in the era of COVID-19 Discussed the role of AI as a technological enabler from four different perspectives in the era of COVID-19
Fong et al. [233]	2020	Literature review	19
Latif et al. [234]	2020	Systematic Review	Discussed many public datasets/repositories that are used in order to track the spread of COVID-19 and mitigation strategies
Chawki et al. [235]	2021	Systematic review	Discussed about how AI can be utilized to analyze the social and clinical patterns of a COVID-19 outbreak to save people
Gunasekeran et al. [236]	2021	Scoping review	Discussed many applications of AI, telehealth, and relevant digital health solutions amidst the COVID- 19
Syeda et al. [237]	2021	Systematic review	Discussed many studies concerning COVID-19 that have utilized AI-based methods in different themes
Zhao et al. [238]	2021	Systematic review	Discussed and summarized 50 applications of AI, robotics, and other digital technologies in the era of COVID-19
Kamalov et al. [239]	2021	Literature review	Discussed AI applications from four perspectives such as medical diagnostics, forecasting, contact tracing, and drug development
Safdari et al. [240]	2021	Scoping review	Discussed about determining the most favorite and effective data mining tools in COVID-19 era
Rasheed et al. [242]	2021	Literature review	Discussed the role of AI from three perspectives such as analyze, prognosis, and tracking of the COVID-19 cases
Senthilraja et al. [243]	2021	Literature review	Discussed and find that AI is useful not only in treatment of infected patients with COVID-19, but also for proper health monitoring
Kumar et al. [244]	2021	Literature review	Discussed the development of COVID-19 classification tools & drug discovery models for infected patients using AI
Chen et al. [245]	2021	Literature survey	Investigated the scope of AI in COVID-19 era from the five aspects (i.e., virology, diagnosis, drug analysis, and transmission)
Dogan et al. [246]	2021	Systematic review	Analyzed the role of AI/ML for transmission prediction, diagnosis, and drug/vaccine development in the pandemic arena
Alafif et al. [247]	2021	Systematic review	Analyzed the role of ML/DL towards COVID-19 diagnosis and treatment and discussed findings of SOTA in the pandemic arena
Singh et al. [248]	2021	Comprehensive review	Discussed about the COVID-19 prevention and detection using different types of biosensors.

Recently, AI has been widely used in COVID-19 drug design and repurposing on different datasets. Tang et al. [249] used AI techniques to predict molecules and leading compounds for each target. The dataset used in the study is available at the link (https://github.com/tbwxmu/2019-nCov (accessed on 15 November 2021)). Similarly, some other studies have also used real-world datasets to find the molecular structures for 3CLpro [250,251]. The data used in these studies can be found at link (https://www.insilico.com/ncov-sprint (accessed on 15 November 2021), https://github.com/ml-jku/sarscov-inhibitors-chemai (accessed on 15 November 2021)). More details about datasets used in drug design can be learned from Chen et al. [245].

#### 5.4. Future Research Directions

The promising research directions that need further exploration from the research and development point of view are described in Figure 16. Besides the other directions, privacy preservation is one of the hot research topics in the pandemic era [252,253].



Figure 16. Promising research directions in the COVID-19 era leveraging AI.

As shown in Figure 16, fusion of heterogeneous sources data for insight (i.e., contacts of an infected individual, stay points of an individual, and co-relation of the symptoms with underlying diseases, etc.) finding is a promising avenue for research in the near future [254]. During the pandemic, there is an emerging need to secure all phases of the data lifecycle in order to prevent privacy breaches [255]. However, the existing approaches mainly ensure security of one/two phases. Therefore, more practical and robust AI-powered privacy preserving approaches are needed to secure personal data in the post COVID-19 era. In the ongoing pandemic, many AI models have been used for multiple purposes. Moreover, designing low-cost AI models and metrics for COVID-19-like pandemics is an emerging avenue for future research [256]. Apart from the potential avenues of research cited above, performing analytics on the collected data in order to extract insights is a promising area of research due to huge data collection in the ongoing pandemic [257,258]. As the healthcare industry is aiming to shift from the hospital-centered approach to patient/device-centered approach, therefore, AI-based methods for supporting the cause are needed in the near future. Furthermore, exploring the potential of other latest technologies (i.e., blockchain, privacy by design, federated learning, swarm learning, few-short learning, deep/machine learning, etc.) to serve mankind in an effective way compared to the recent past is an emerging avenue of the research. Finally, critical analysis of AI use from ethical point of view in the context of ongoing pandemic has become more emergent than ever [259-261]. Besides the AI applications cited above, joint use of AI with the IoT/IoMT has become a popular research area [262–265]. Recently, researchers have started working on reducing the 'blackbox' nature of AI models through explain-ability and interpret-ability concepts [266–269]. Hence, it is worth exploring the AI use in COVID-19 context from this perspective [270]. In addition, most of the AI techniques, especially deep learning techniques, are computationally expensive. Hence, it is an emerging research area to lower the computing burden of these techniques via pruning and quantization techniques [271–274]. Besides the areas cited above, another promising avenue for future research is sentiment analysis of COVID-19-related tweets [275], informative tweets detection related to COVID-19 using deep learning [276], reviews analysis [277], topic modeling related to COVID-19 aspects [278], COVID-19 pandemic and vaccine-related rumors detection [279], opinion analysis related to COVID-19 [280], and awareness prediction [281], to name a few. In these areas, AI can play a vital role with data stemming from the ongoing pandemic and corresponding surge in SN use across the globe. Therefore, further research and development are likely to expand in this regard (e.g., AI towards COVID-19) in order to take full advantage of the integrated technologies to serve humanity.

Lastly, the COVID-19 pandemic has evolved into an endemic, and the new normal may be to live with the virus for a few more years. Therefore, we must recognize and advocate caution that such pre-emptive measures (e.g., heavy reliance on the digital tools,

contactless services, work from home, remote health monitoring with AI tools/applications, distance/remote learning [282], and buying online, to name a few) are ultimately worthwhile. Considering the evolution of COVID-19 in various mutations across the globe, preparedness is the key to preventing any public health crisis, and COVID-19-endemic countries must be ready for the challenges that COVID-19 might bring unanticipated strain on medical infrastructure time and again. Furthermore, some tropical diseases (TDs) have been neglected amid the prevalence of COVID-19 because most funds are being redirected towards COVID-19 [283]. Therefore, COVID-19 effects on medical systems are long-lasting. The contactless services are increasing significantly amid the ongoing pandemic, and hybrid healthcare services will likely increase in the post-COVID-19 era [284]. Most companies are moving towards zero user interface (ZUI) technologies propelled by the ongoing pandemic in order to meet the hygiene requirements [285]. The cashless and contactless smart vending machines are being integrated with the mobile device to reduce people-to-people contact [286]. Furthermore, new normal activities are constantly emerging in which companies are deploying crisis strategies to retain their stakeholders and business. Companies are re-opening swiftly, focusing more on digital transformations, implementing digital platforms/tools for improved consumer services and ease in working, to name a few operational changes [287]. Besides these technical developments, researchers are emphasizing the need for research in COVID19-induced brain dysfunction (CIBD) to improve the mental health of large populations of infected /uninfected individuals [288]. Despite these highlighted areas and developments, AI's role in COVID-19 and post COVID-19 era can offer a proven method to further strengthen the impact of healthcare services/applications on population health, which is more necessary than ever in the post-COVID era [289–292]. To this end, further developments are needed to lower the effects of this pandemic and to serve mankind effectively in these unanticipated and challenging times.

#### 6. Conclusions and Future Work

This paper has demonstrated the role of data-driven analytics leveraging AI in the era of COVID-19. Specifically, we have discussed the role of AI from six different perspectives in the pandemic arena that can assist early researchers to grasp the research status conveniently. To the best of author knowledge, this is the first article that has presented various possible and demonstrated applications of Artificial Intelligence (AI) related to the COVID-19 pandemic. It continues to discuss challenges facing the field and proposes future avenues of research to follow. The six unique perspectives in which AI's use/role was presented are, (i) AI role in seven different epidemic containment strategies (a.k.a non-pharmaceutical interventions (NPIs)) such as contact tracing, quarantine monitoring, social distance monitoring, disclosing patients' information, reporting symptoms and other data via wearable devices, data collection at the time of check-ups, and mining and analytics on collected data, (ii) AI role in data life cycle phases (i.e., collection, storage, pre-processing, analytics, use, distribution, archiving, and deletion) employed for epidemic handing in digital solutions, (iii) AI role in performing analytics on heterogeneous types of data stemming from this pandemic, (iv) AI role in the healthcare sector in the context of COVID-19 pandemic, (v) general-purpose applications of AI in COVID-19 era, and (vi) AI role in drug design and repurposing against COVID-19. Furthermore, we discussed the various challenges related to data that can hinder the application of AI in the ongoing pandemic period. We have concisely presented the role of emerging technologies other than AI, and promising future research directions in the post-COVID-19 era. Furthermore, we have demonstrated the actual use of AI based on available data and utility in these pandemic times. We have described the findings of recently published SOTA studies that have demonstrated the role of IoT and various smart sensing technologies in order to fight against the COVID-19 pandemic. Furthermore, we discussed the synergy of AI with other emerging technologies in order to lower the effects of COVID-19 on the general public and economies. With this comprehensive overview, we aim to update researchers and developers with the existing services of AI, and possible research gaps/opportunities that AI can provide in the near future, and data-related issues that we may face while applying

AI models on it. Unfortunately, COVID-19 is creating many complex clinical implications for people having underlying diseases such as diabetes, pneumonia, and heart disease, to name a few. Hence, we may need to deal with the clinical implications emerging from the prognosis of COVID-19 using AI, mastering the "fearful symmetry" [293]. In the future, we aim to explore the development-related challenges of AI in the COVID-19 era. Finally, we intend to explore AI's role and applications in the post-pandemic era. Recently, federated analytics has emerged as a new paradigm for performing analytics without centralizing data [294–296]. To this end, we intend to explore the role of federated analytics in the pandemic and post-pandemic arena. In addition, studying the role of federated learning in the context of COVID-19 is also a very hot research area in recent times [297–299]. We intend to explore the role of these emerging technologies to combat this unanticipated pandemic in future work.

**Author Contributions:** All authors contributed equally to this work. All authors have read and agreed to the published version of the manuscript.

**Funding:** This research was supported by the MSIT (Ministry of Science, ICT), Korea, under the High-Potential Individuals Global Training Program (No. 2021-0-01532, 50%) supervised by the IITP (Institute for Information & Communications Technology Planning & Evaluation), and National Research Foundation of Korea (NRF) grant (No.2020R1A2B5B01002145, 50%).

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

**Conflicts of Interest:** The authors declare no conflict of interest.

#### References

- 1. Alzween, F.F. Social Distancing and its Relationship to Psychological Stress among a Sample of Saudis during COVID-19 Pandemic. *Psychol. Educ. J.* **2021**, *58*, 3208–3222.
- Ribeiro, F.; Leist, A. Who is going to pay the price of Covid-19? Reflections about an unequal Brazil. *Int. J. Equity Health* 2020, 19, 1–3. [CrossRef] [PubMed]
- 3. Gunawan, J.; Juthamanee, S.; Aungsuroch, Y. Current mental health issues in the era of Covid-19. *Asian J. Psychiatry* **2020**, *51*, 102103. [CrossRef]
- de Sá, A.A.; Carvalho, J.D.; Naves, E.L. Reflections on epistemological aspects of artificial intelligence during the COVID-19 pandemic. *AI Soc.* 2021, 1–8. [CrossRef] [PubMed]
- 5. Wang, W.; Siau, K. Artificial intelligence, machine learning, automation, robotics, future of work and future of humanity: A review and research agenda. J. Database Manag. (JDM) 2019, 30, 61–79. [CrossRef]
- 6. Garg, R.; Patel, A.; Hoda, W. Emerging role of artificial intelligence in medical sciences—Are we ready! *J. Anaesthesiol. Clin. Pharmacol.* **2021**, *37*, 35. [CrossRef]
- Shaw, R.; Kim, Y.-K.; Hua, J. Governance, technology and citizen behavior in pandemic: Lessons from COVID-19 in East Asia. Prog. Disaster Sci. 2020, 6, 100090. [CrossRef]
- 8. Mbunge, E.; Akinnuwesi, B.; Fashoto, S.G.; Metfula, A.S.; Mashwama, P. A critical review of emerging technologies for tackling COVID-19 pandemic. *Hum. Behav. Emerg. Technol.* **2021**, *3*, 25–39. [CrossRef]
- 9. Allam, Z.; Dey, G.; Jones, D.S. Artificial intelligence (AI) provided early detection of the coronavirus (COVID-19) in China and will influence future Urban health policy internationally. *AI* **2020**, *1*, 9. [CrossRef]
- 10. Liu, J.; Wang, Z.; Huang, S.; Ren, A. An Overview of Healthcare Information Technologies Used to Combat the COVID-19 Pandemic in China. *Int. J. Digit. Health* **2021**, *1*, 14. [CrossRef]
- 11. Sun, S.; Xie, Z.; Yu, K.; Jiang, B.; Zheng, S.; Pan, X. COVID-19 and healthcare system in China: Challenges and progression for a sustainable future. *Glob. Health* **2021**, *17*, 1–8. [CrossRef]
- 12. Gao, X.; Xu, J.; Liu, H. Current Status of Healthcare and Available E-Health Solutions in China. In *E-Business in the 21st Century: Essential Topics and Studies*; World Scientific Pub. Co. Inc.: Singapore, 2021; pp. 169–199.
- 13. Mathe, N. Insights from' Unlocking COVID-19 current realities, future opportunities: Artificial intelligence in the time of COVID-19'. S. Afr. J. Sci. 2021, 117, 1–2. [CrossRef]
- 14. Wang, X.V.; Wang, L. A literature survey of the robotic technologies during the COVID-19 pandemic. *J. Manuf. Syst.* 2021, 60, 823–836. [CrossRef] [PubMed]
- 15. Qi, Q.; Tao, F.; Cheng, Y.; Cheng, J.; Nee, A.Y.C. New IT driven rapid manufacturing for emergency response. *J. Manuf. Syst.* 2021, 60, 928–935. [CrossRef] [PubMed]
- 16. Khemasuwan, D.; Colt, H.G. Applications and challenges of AI-based algorithms in the COVID-19 pandemic. *BMJ Innov.* **2021**, *7*, 387–398. [CrossRef]

- 17. Choi, T.M. Fighting against COVID-19: What operations research can help and the sense-and-respond framework. *Ann. Oper. Res.* **2021**, *5*, 1–17. [CrossRef]
- Raza, K.; Qazi, S. An introduction to computational intelligence in COVID-19: Surveillance, prevention, prediction, and diagnosis. In *Computational Intelligence Methods in COVID-19: Surveillance, Prevention, Prediction and Diagnosis*; Springer: Singapore, 2021; pp. 3–18.
- 19. Tariq, A.; Celi, L.A.; Newsome, J.M.; Purkayastha, S.; Bhatia, N.K.; Trivedi, H.; Gichoya, J.W.; Banerjee, I. Patient-specific COVID-19 resource utilization prediction using fusion AI model. *NPJ Digit. Med.* **2021**, *4*, 1–9. [CrossRef]
- Snider, B.; McBean, E.A.; Yawney, J.; Gadsden, S.A.; Patel, B. Identification of Variable Importance for Predictions of Mortality from COVID-19 Using AI Models for Ontario, Canada. *Front. Public Health* 2021, 9, 675766. [CrossRef]
- 21. Mangold, C.; Zoretic, S.; Thallapureddy, K.; Moreira, A.; Chorath, K.; Moreira, A. Machine Learning Models for Predicting Neonatal Mortality: A Systematic Review. *Neonatology* **2021**, *118*, 394–405. [CrossRef]
- Casaccia, S.; Revel, G.M.; Cosoli, G.; Scalise, L. Assessment of domestic well-being: from perception to measurement. *IEEE Instrum. Meas. Mag.* 2021, 24, 58–67. [CrossRef]
- 23. Trocin, C.; Mikalef, P.; Papamitsiou, Z.; Conboy, K. Responsible AI for digital health: A synthesis and a research agenda. *Inf. Syst. Front.* **2021**, 1–19. [CrossRef]
- Fang, X.; Kruger, U.; Homayounieh, F.; Chao, H.; Zhang, J.; Digumarthy, S.R.; Arru, C.D.; Kalra, M.K.; Yan, P. Association of AI quantified COVID-19 chest CT and patient outcome. *Int. J. Comput. Assist. Radiol. Surg.* 2021, *16*, 435–445. [CrossRef]
- 25. Arora, N.; Banerjee A.K.; Narasu, M.L. The role of artificial intelligence in tackling COVID-19. *Future Virol.* **2020**, 717–724. [CrossRef]
- Huang, S.; Yang, J.; Fong, S.; Zhao, Q. Artificial intelligence in the diagnosis of COVID-19: Challenges and perspectives. *Int. J. Biol. Sci.* 2021, 17, 1581. [CrossRef] [PubMed]
- 27. da Motta, O.J.R.; Machado, G.R.; Gomes, A.P.; Nas, E.; Silva, E.; Goldschmidt, R.R.; Siqueira-Batista, R. COVID-19 Pandemic: How Artificial Intelligence can help us. *Braz. Appl. Sci. Rev.* **2020**, *4*, 2904–2915. [CrossRef]
- 28. Cave, S.; Whittlestone, J.; Nyrup, R.; Calvo, R.A. Using AI ethically to tackle COVID-19. BMJ 2021, 372, n364. [CrossRef] [PubMed]
- Leslie, D.; Mazumder, A.; Peppin, A.; Wolters, M.K.; Hagerty, A. Does "AI" stand for augmenting inequality in the era of COVID-19 healthcare? *BMJ* 2021, 372, n304. [CrossRef]
- 30. The Lancet Digital Health. Artificial Intelligence for COVID-19: Saviour or Saboteur? Lancet Digit. Health 2021, 3, e1. [CrossRef]
- 31. Chang, A.C. Artificial intelligence and COVID-19: Present state and future vision. Intell.-Based Med. 2020, 3, 100012. [CrossRef]
- Vaishya, R.; Javaid, M.; Khan, I.H.; Haleem, A. Artificial Intelligence (AI) applications for COVID-19 pandemic. *Diabetes Metab.* Syndr. Clin. Res. Rev. 2020, 14, 337–339. [CrossRef]
- Surianarayanan, C.; Chelliah, P.R. Leveraging Artificial Intelligence (AI) Capabilities for COVID-19 Containment. New Gener. Comput. 2021, 39, 717–741. [CrossRef] [PubMed]
- Abdul Salam, M.; Taha, S.; Ramadan, M. COVID-19 detection using federated machine learning. *PLoS ONE* 2021, 16, e0252573. [CrossRef] [PubMed]
- 35. Soomro, T.A.; Zheng, L.; Afifi, A.J.; Ali, A.; Yin, M.; Gao, J. Artificial intelligence (AI) for medical imaging to combat coronavirus disease (COVID-19): A detailed review with direction for future research. *Artif. Intell. Rev.* 2021, *15*, 1–31. [CrossRef] [PubMed]
- Payedimarri, A.B.; Concina, D.; Portinale, L.; Canonico, M.; Seys, D.; Vanhaecht, K.; Panella, M. Prediction Models for Public Health Containment Measures on COVID-19 Using Artificial Intelligence and Machine Learning: A Systematic Review. *Int. J. Environ. Res. Public Health* 2021, 18, 4499. [CrossRef]
- Sakib, S.; Fouda, M.M.; Fadlullah, Z.M.; Nasser, N. On COVID-19 prediction using asynchronous federated learning-based agile radiograph screening booths. In Proceedings of the ICC 2021-IEEE International Conference on Communications, Montreal, QC, Canada, 14–23 June 2021; pp. 1–6.
- Vecino-Ortiz, A.I.; Congote, J.V.; Bedoya, S.Z.; Cucunuba, Z.M. Impact of contact tracing on COVID-19 mortality: An impact evaluation using surveillance data from Colombia. *PLoS ONE* 2021, 16, e0246987. [CrossRef]
- 39. Hsu, J. The Dilemma of contacttracing apps: Can this crucial technology be both effective and private? *IEEE Spectr.* **2020**, *57*, 56–59. [CrossRef]
- Szocska, M.; Pollner, P.; Schiszler, I.; Joo, T.; Palicz, T.; McKee, M.; Asztalos, A.; Bencze, L.; Kapronczay, M.; Petrecz, P.; et al. Countrywide population movement monitoring using mobile devices generated (big) data during the COVID-19 crisis. *Sci. Rep.* 2021, 11, 1–9. [CrossRef]
- 41. Arellano, J.R.; Delgado, J.E. Lived Experiences of Persons under Investigation and Persons under Monitoring During General Community Quarantine. *South Asian J. Soc. Stud. Econ.* **2021**, 20–26. [CrossRef]
- Ahmed, I.; Ahmad, M.; Rodrigues, J.J.; Jeon, G.; Din, S. A deep learning-based social distance monitoring framework for COVID-19. Sustain. Cities Soc. 2021, 65, 102571. [CrossRef]
- 43. Ahmed, I.; Ahmad, M.; Jeon, G. Social distance monitoring framework using deep learning architecture to control infection transmission of COVID-19 pandemic. *Sustain. Cities Soc.* **2021**, *69*, 102777. [CrossRef]
- Kalra, J.S.; Pant, R.; Gupta, S.; Kumar, V. Social Distance Monitoring in Smart Cities using IoT. In *Green Internet of Things for Smart Cities*; CRC Press: Boca Raton, FL, USA, 2021; pp. 135–145.
- Narvaez, A.A.; Guerra, J.G. Received signal strength indication—Based COVID-19 mobile application to comply with social distancing using bluetooth signals from smartphones. In *Data Science for COVID-19*; Academic Press: Cambridge, MA, USA, 2021; pp. 483–501.

- Molldrem, S.; Hussain, M.I.; McClelland, A. Alternatives to sharing COVID-19 data with law enforcement: Recommendations for stakeholders. *Health Policy* 2021, 125, 135–140. [CrossRef] [PubMed]
- Clingan, C.A.; Dittakavi, M.; Rozwadowski, M.; Gilley, K.N.; Cislo, C.R.; Barabas, J.; Sandford, E.; Olesnavich, M.; Flora, C.; Tyler, J.; et al. Monitoring Health Care Workers at Risk for COVID-19 Using Wearable Sensors and Smartphone Technology: Protocol for an Observational mHealth Study. *JMIR Res. Protoc.* 2021, 10, e29562. [CrossRef] [PubMed]
- 48. Quer, G.; Radin, J.M.; Gadaleta, M.; Baca-Motes, K.; Ariniello, L.; Ramos, E.; Kheterpal, V.; Topol, E.J.; Steinhubl, S.R. Wearable sensor data and self-reported symptoms for COVID-19 detection. *Nat. Med.* **2021**, *27*, 73–77. [CrossRef] [PubMed]
- Rehman, A.; Saba, T.; Tariq, U.; Ayesha, N. Deep learning-based COVID-19 detection using CT and X-ray images: Current analytics and comparisons. *IT Prof.* 2021, 23, 63–68. [CrossRef]
- 50. Ebadi, A.; Xi, P.; MacLean, A.; Tremblay, S.; Kohli, S.; Wong, A. COVIDx-US–An open-access benchmark dataset of ultrasound imaging data for AI-driven COVID-19 analytics. *arXiv* 2021, arXiv:2103.10003.
- Praveen, S.V.; Ittamalla, R.; Deepak, G. Analyzing the attitude of Indian citizens towards COVID-19 vaccine—A text analytics study. Diabetes Metab. Syndr. Clin. Res. Rev. 2021, 15, 595–599. [CrossRef] [PubMed]
- Park, Y.-E. Developing a COVID-19 Crisis Management Strategy Using News Media and Social Media in Big Data Analytics. Soc. Sci. Comput. Rev. 2021, 08944393211007314. [CrossRef]
- 53. Morris, K. Smart Biomedical Sensors, Big Healthcare Data Analytics, and Virtual Care Technologies in Monitoring, Detection, and Prevention of COVID-19. *Am. J. Med Res.* **2021**, *8*, 60–70.
- Alsunaidi, S.J.; Almuhaideb, A.M.; Ibrahim, N.M.; Shaikh, F.S.; Alqudaihi, K.S.; Alhaidari, F.A.; Khan, I.U.; Aslam, N.; Alshahrani, M.S. Applications of Big Data Analytics to Control COVID-19 Pandemic. *Sensors* 2021, 21, 2282. [CrossRef]
- 55. Jung, G.; Lee, H.; Kim, A.; Lee, U. Too much information: Assessing privacy risks of contact trace data disclosure on people with COVID-19 in South Korea. *Front. Public Health* **2020**, *8*, 305. [CrossRef]
- 56. Jaswal, G.; Bharadwaj, R.; Tiwari, K.; Thapar, D.; Goyal, P.; Nigam, A. AI-Biometric-Driven Smartphone App for Strict Post-COVID Home Quarantine Management. *IEEE Consum. Electron. Mag.* **2020**, *10*, 49–55. [CrossRef]
- Jain, A.; Kushwah, R.; Swaroop, A.; Yadav, A. Role of Artificial Intelligence of Things (AIoT) to Combat Pandemic COVID-19. In *Handbook of Research on Innovations and Applications of AI, IoT, and Cognitive Technologies*; IGI Global: Hershey, PA, USA, 2021; pp. 117–128.
- 58. Ahn, P. D.; Wickramasinghe, D. Pushing the limits of accountability: Big data analytics containing and controlling COVID-19 in South Korea. *Account. Audit. Account. J.* 2021. [CrossRef]
- 59. Crossley, T.F.; Fisher, P.; Low, H. The heterogeneous and regressive consequences of COVID-19: Evidence from high quality panel data. *J. Public Econ.* **2021**, *193*, 104334. [CrossRef] [PubMed]
- 60. Azeroual, O.; Fabre, R. Processing Big Data with Apache Hadoop in the Current Challenging Era of COVID-19. *Big Data Cogn. Comput.* **2021**, *5*, 12. [CrossRef]
- 61. Park, S.; Choi, G.J.; Ko, H. Privacy in the Time of COVID-19: Divergent Paths for Contact Tracing and Route-Disclosure Mechanisms in South Korea. *IEEE Secur. Priv.* 2021, *19*, 51–56. [CrossRef]
- Kim, J.; Kwan, M.P. An examination of people's privacy concerns, perceptions of social benefits, and acceptance of COVID-19 mitigation measures that harness location information: A comparative study of the US and South Korea. *ISPRS Int. J. Geo-Inf.* 2021, 10, 25. [CrossRef]
- 63. Ahmad, N.; Chauhan, P. State of Data Privacy During COVID-19. IEEE Ann. Hist. Comput. 2020, 53, 119–122. [CrossRef]
- Hensen, B.; Mackworth-Young, C.R.S.; Simwinga, M.; Abdelmagid, N.; Banda, J.; Mavodza, C.; Doyle, A.M.; Bonell, C.; Weiss, H.A. Remote data collection for public health research in a COVID-19 era: Ethical implications, challenges and opportunities. *Health Policy Plan.* 2021, *36*, 360–368. [CrossRef] [PubMed]
- Gadaleta, M.; Radin, J.M.; Baca-Motes, K.; Ramos, E.; Kheterpal, V.; Topol, E.J.; Steinhubl, S.R.; Quer, G. Passive detection of COVID-19 with wearable sensors and explainable machine learning algorithms. *Npj Digit. Med.* 2021, 4, 1–10. [CrossRef] [PubMed]
- 66. Belliveau, R. The Information Infrastructure of Search Engines: A Content Analysis of Covid-19 Immunization Information Online. J. Grad. Stud. J. Fac. Inf. 2021, 6, 1-9. [CrossRef]
- 67. Fiala, T. Commentary on: Data-Driven Insights on the Effects of COVID-19 (Parts I and II). *Aesthetic Surg. J.* **2021**, *41*, NP83–NP87. [CrossRef] [PubMed]
- 68. Kuppalli, K.; Gala, P.; Cherabuddi, K.; Kalantri, S.P.; Mohanan, M.; Mukherjee, B.; Pinto, L.; Prakash, M.; Pramesh, C.S.; Rathi, S.; et al. India's COVID-19 crisis: A call for international action. *Lancet* **2021**, 397, 2132–2135. [CrossRef]
- Megreya, A.M.; Latzman, R.D.; Al-Ahmadi, A.M.; Al-Dosari, N.F. The COVID-19-Related Lockdown in Qatar: Associations Among Demographics, Social Distancing, Mood Changes, and Quality of Life. Int. J. Ment. Health Addict. 2021, 1–17. [CrossRef] [PubMed]
- Mollalo, A.; Rivera, K.M.; Vahedi, B. Artificial neural network modeling of novel coronavirus (COVID-19) incidence rates across the continental United States. *Int. J. Environ. Res. Public Health* 2020, 17, 4204. [CrossRef] [PubMed]
- Asada, K.; Komatsu, M.; Shimoyama, R.; Takasawa, K.; Shinkai, N.; Sakai, A.; Bolatkan, A.; Yamada, M.; Takahashi, S.; Machino, H.; et al. Application of Artificial Intelligence in COVID-19 Diagnosis and Therapeutics. *J. Pers. Med.* 2021, *11*, 886. [CrossRef] [PubMed]
- 72. Wang, B.; Jin, S.; Yan, Q.; Xu, H.; Luo, C.; Wei, L.; Zhao, W.; Hou, X.; Ma, W.; Xu, Z.; et al. AI-assisted CT imaging analysis for COVID-19 screening: Building and deploying a medical AI system. *Appl. Soft Comput.* **2021**, *98*, 106897. [CrossRef] [PubMed]

- 73. Feng, S.; Feng, Z.; Ling, C.; Chang, C.; Feng, Z. Prediction of the COVID-19 epidemic trends based on SEIR and AI models. *PLoS ONE* **2021**, *16*, e0245101. [CrossRef] [PubMed]
- Abdel-Basset, M.; Chang, V.; Nabeeh, N.A. An intelligent framework using disruptive technologies for COVID-19 analysis. *Technol. Forecast. Soc. Chang.* 2021, 163, 120431. [CrossRef] [PubMed]
- Qian, K.; Zhang, Z.; Yamamoto, Y.; Schuller, B.W. Artificial intelligence internet of things for the elderly: From assisted living to health-care monitoring. *IEEE Signal Process. Mag.* 2021, 38, 78–88. [CrossRef]
- Saxena, A.; Chandra, S. Visualization and Prediction of COVID-19 Using AI and ML. In Artificial Intelligence and Machine Learning in Healthcare; Springer: Singapore, 2021; pp. 99–112.
- 77. Haqbeen, J.; Ito, T.; Sahab, S.; Hadfi, R.; Okuhara, S.; Saba, N.; Hofaini, M.; Baregzai, U. A contribution to COVID-19 prevention through crowd collaboration using conversational AI & social platforms. *arXiv* **2021**, arXiv:2106.11023.
- Tilahun, B.; Gashu, K.D.; Mekonnen, Z.A.; Endehabtu, B.F.; Angaw, D.A. Mapping the Role of Digital Health Technologies in Prevention and Control of COVID-19 Pandemic: Review of the Literature. *Yearb. Med Inform.* 2021, 30, 26–37. [CrossRef] [PubMed]
- 79. Migdadi, Y.K.A.-A. Airline effective operations strategy during COVID-19 pandemic: Across regional worldwide survey. *Rev. Int. Bus. Strategy* **2021**. [CrossRef]
- Abeyrathna, K.D.; Rasca, S.; Markvica, K.; Granmo, O.-C. Public Transport Passenger Count Forecasting in Pandemic Scenarios Using Regression Tsetlin Machine. Case Study of Agder, Norway. In *Smart Transportation Systems*; Springer: Singapore, 2021; pp. 27–37.
- 81. Xiang, S.; Rasool, S.; Hang, Y.; Javid, K.; Javed, T.; Artene, A.E. The Effect of COVID-19 Pandemic on Service Sector Sustainability and Growth. *Front. Psychol.* **2021**, *12*, 633597. [CrossRef]
- García-Cremades, S.; Morales-García, J.; Hernández-Sanjaime, R.; Martínez-España, R.; Bueno-Crespo, A.; Hernández-Orallo, E.; López-Espín, J.J.; Cecilia, J.M. Improving prediction of COVID-19 evolution by fusing epidemiological and mobility data. *Sci. Rep.* 2021, 11, 1–16. [CrossRef]
- Jahja, M.; Chin, A.; Tibshirani, R.J. Real-Time Estimation of COVID-19 Infections via Deconvolution and Sensor Fusion. *arXiv* 2021, arXiv:2112.06697.
- 84. Shin, K.-Y. Work in the post-COVID-19 pandemic: The case of South Korea. Globalizations 2021, 1–10. [CrossRef]
- 85. Ning, J.; Chu, Y.; Liu, X.; Zhang, D.; Zhang, J.; Li, W.; Zhang, H. Spatio-temporal characteristics and control strategies in the early period of COVID-19 spread: A case study of the mainland China. *Environ. Sci. Pollut. Res.* 2021, 28, 48298–48311. [CrossRef]
- 86. Selvaraj, C.; Chandra, I.; Singh, S.K. Artificial intelligence and machine learning approaches for drug design: Challenges and opportunities for the pharmaceutical industries. *Mol. Divers.* **2021**, 1–21. [CrossRef] [PubMed]
- 87. Arora, G.; Joshi, J.; Mandal, R.S.; Shrivastava, N.; Virmani, R.; Sethi, T. Artificial Intelligence in Surveillance, Diagnosis, Drug Discovery and Vaccine Development against COVID-19. *Pathogens* **2021**, *10*, 1048. [CrossRef] [PubMed]
- 88. Monteleone, S.; Kellici, T.F.; Southey, M.; Bodkin, M.J.; Heifetz, A. Fighting COVID-19 with Artificial Intelligence. In *Artificial Intelligence in Drug Design. Methods in Molecular Biology*; Heifetz, A., Ed.; Humana: New York, NY, USA, 2022; Volume 2390.
- 89. Zhou, Y.; Wang, F.; Tang, J.; Nussinov, R.; Cheng, F. Artificial intelligence in COVID-19 drug repurposing. *Lancet Digit. Health* 2020, 2, e667–e676. [CrossRef]
- 90. Walters, W.P.; Barzilay, R. Applications of deep learning in molecule generation and molecular property prediction. *Accounts Chem. Res.* 2020, 54, 263–270. [CrossRef]
- 91. Patronov, A.; Papadopoulos, K.; Engkvist, O. Has Artificial Intelligence Impacted Drug Discovery? In *Artificial Intelligence in Drug Design*; Humana: New York, NY, USA, 2022; pp. 153–176.
- 92. Arora, K.; Bist, A.S. Artificial intelligence based drug discovery techniques for COVID-19 detection. *Aptisi Trans. Technopreneurship* (*ATT*) **2020**, *2*, 120–126. [CrossRef]
- Bhati, A.P.; Wan, S.; Alfè, D.; Clyde, A.R.; Bode, M.; Tan, L.; Titov, M.; Merzky, A.; Turilli, M.; Jha, S.; et al. Pandemic drugs at pandemic speed: Infrastructure for accelerating COVID-19 drug discovery with hybrid machine learning-and physics-based simulations on high-performance computers. *Interface Focus* 2021, *11*, 20210018. [CrossRef]
- Kabra, R.; Singh, S. Evolutionary artificial intelligence based peptide discoveries for effective Covid-19 therapeutics. *Biochim. Biophys. Acta (BBA) Mol. Basis Dis.* 2021, 1867, 165978. [CrossRef] [PubMed]
- 95. Bai, Q.; Tan, S.; Xu, T.; Liu, H.; Huang, J.; Yao, Z. MolAICal: A soft tool for 3D drug design of protein targets by artificial intelligence and classical algorithm. *Briefings Bioinform.* **2021**, *22*, bbaa161. [CrossRef]
- 96. Liu, H.; Lin, H.; Shen, C.; Yang, L.; Lin, Y.; Xu, B.; Yang, Z.; Wang, J.; Sun, Y. A network representation approach for COVID-19 drug recommendation. *Methods* **2021**. [CrossRef] [PubMed]
- Delijewski, M.; Haneczok, J. AI drug discovery screening for COVID-19 reveals zafirlukast as a repurposing candidate. *Med. Drug Discov.* 2021, 9, 100077. [CrossRef] [PubMed]
- 98. Haneczok, J.; Delijewski, M. Machine learning enabled identification of potential SARS-CoV-2 3CLpro inhibitors based on fixed molecular fingerprints and Graph-CNN neural representations. *J. Biomed. Inform.* **2021**, *119*, 103821. [CrossRef]
- 99. Chen, R.J.; Lu, M.Y.; Chen, T.Y.; Williamson, D.F.K.; Mahmood, F. Synthetic data in machine learning for medicine and healthcare. *Nat. Biomed. Eng.* **2021**, *5*, 493–497 [CrossRef]
- Islam, M.N.; Inan, T.T.; Rafi, S.; Akter, S.S.; Sarker, I.H.; Islam, A.K.M.R. A Systematic Review on the Use of AI and ML for Fighting the COVID-19 Pandemic. *IEEE Trans. Artif. Intell.* 2021, 1, 258–270. [CrossRef]

- 101. Shamout, F.E.; Shen, Y.; Wu, N.; Kaku, A.; Park, J.; Makino, T.; Jastrzębski, S.; Witowski, J.; Wang, D.; Zhang, B.; et al. An artificial intelligence system for predicting the deterioration of COVID-19 patients in the emergency department. NPJ Digit. Med. 2021, 4, 1–11. [CrossRef]
- 102. Piccialli, F.; di Cola, V.S.; Giampaolo, F.; Cuomo, S. The Role of Artificial Intelligence in Fighting the COVID-19 Pandemic. *Inf. Syst. Front.* **2021**, *23*, 1467–1497. [CrossRef]
- Thomas, S.; Abraham, A.; Baldwin, J.; Piplani, S.; Petrovsky, N.. Artificial Intelligence in Vaccine and Drug Design. In *Vaccine Design*; Humana: New York, NY, USA, 2022; pp. 131–146.
- Mukhtar, A.H.; Hamdan, A. Artificial Intelligence and Coronavirus COVID-19: Applications, Impact and Future Implications. Lect. Notes Netw. Syst. 2021, 194, 830–843.
- 105. Ahuja, V.; Nair, L.V. Artificial Intelligence and technology in COVID Era: A narrative review. *J. Anaesthesiol. Clin. Pharmacol.* **2021**, 37, 28. [CrossRef] [PubMed]
- Pinter, G.; Felde, I.; Mosavi, A.; Ghamisi, P.; Gloaguen, R. COVID-19 Pandemic Prediction for Hungary; A Hybrid Machine Learning Approach. *Mathematics* 2020, *8*, 890. [CrossRef]
- Aminu, M.; Ahmad, N.A.; Noor, M.H.M. Covid-19 detection via deep neural network and occlusion sensitivity maps. *Alex. Eng.* J. 2021, 60, 4829–4855. [CrossRef]
- Magar, R.; Yadav, P.; Farimani, A.B. Potential neutralizing antibodies discovered for novel corona virus using machine learning. *Sci. Rep.* 2021, 11, 1–11. [CrossRef]
- 109. Zeng, W.; Gautam, A.; Huson, D.H. On the Application of Advanced Machine Learning Methods to Analyze Enhanced, Multimodal Data from Persons Infected with COVID-19. *Computation* **2021**, *9*, 4. [CrossRef]
- 110. Ashraf, I.; Alnumay, W.S.; Ali, R.; Hur, S.; Bashir, A.K.; Zikria, Y.B. Prediction Models for COVID-19 Integrating Age Groups, Gender, and Underlying Conditions. *Comput. Mater. Contin.* **2021**, *67*, 3009–3044. [CrossRef]
- 111. Shah, P.M.; Ullah, F.; Shah, D.; Gani, A.; Maple, C.; Wang, Y.; Shahid, A.; Abrar, M.; ul Islam, S. Deep GRU-CNN model for COVID-19 detection from chest X-rays data. *IEEE Access* 2021. [CrossRef]
- 112. Prakash, P.N.S.; Hariharan, B.; Kaliraj, S.; Siva, R.; Vivek, D. The impact of various policy factors implemented for controlling the spread of COVID-19. *Mater. Today Proc.* **2021**. [CrossRef]
- 113. Rathod, P.; Usoro, A. Benchmarking Machine Learning Approaches to Predict the Uncertainties of Pandemic Disease: An Explorative Study. In Proceedings of the 7th Annual International Conference on Information Technology and Economic Development, Gregory University, Uturu, Nigeria, 16–18 November, 2020.
- Ullah, S.I.; Salam, A.; Ullah, W.; Imad, M. COVID-19 Lung Image Classification Based on Logistic Regression and Support Vector Machine. In *European, Asian, Middle Eastern, North African Conference on Management & Information Systems*; Springer: Cham, Switzerland, 2021; pp. 13–23.
- 115. AlMeshal; Reem, A.K.H. The Impact of COVID-19 on Arabian Gulf Countries Using the Classical Machine Learning Methods. Available online: https://www.mecsj.com/uplode/images/photo/The\_impact\_of\_COVID-19\_on\_Arabian\_Gulf\_countries\_ using\_the\_Classical\_Machine\_Learning\_Methods.pdf (accessed on 25 October 2021).
- Hu, F.; Huang, M.; Sun, J.; Zhang, X.; Liu, J. An analysis model of diagnosis and treatment for COVID-19 pandemic based on medical information fusion. *Inf. Fusion* 2021, 73, 11–21. [CrossRef] [PubMed]
- 117. Rashed, E.A.; Hirata, A. One-Year Lesson: Machine Learning Prediction of COVID-19 Positive Cases with Meteorological Data and Mobility Estimate in Japan. *Int. J. Environ. Res. Public Health* **2021**, *18*, 5736. [CrossRef] [PubMed]
- 118. Singh, D.; Kumar, V.; Kaur, M. Densely connected convolutional networks-based COVID-19 screening model. *Appl. Intell.* **2021**, *51*, 3044–3051. [CrossRef] [PubMed]
- Saverino, A.; Baiardi, P.; Galata, G.; Pedemonte, G.; Vassallo, C.; Pistarini, C. The Challenge of Reorganizing Rehabilitation Services at the Time of COVID-19 Pandemic: A New Digital and Artificial Intelligence Platform to Support Team Work in Planning and Delivering Safe and High Quality Care. Front. Neurol. 2021, 12, 643251. [CrossRef]
- Peddinti, B.; Shaikh, A.; Bhavya, K.R.; Kumar, N. Framework for Real-Time Detection and Identification of possible patients of COVID-19 at public places. *Biomed. Signal Process. Control.* 2021, 68, 102605. [CrossRef]
- 121. Malla, S.; Alphonse, P.J.A. COVID-19 outbreak: An ensemble pre-trained deep learning model for detecting informative tweets. *Appl. Soft Comput.* **2021**, *107*, 107495. [CrossRef]
- 122. Lella, K.K.; Pja, A. Automatic COVID-19 disease diagnosis using 1D convolutional neural network and augmentation with human respiratory sound based on parameters: Cough, breath, and voice. *AIMS Public Health* **2021**, *8*, 240. [CrossRef]
- 123. Haleem, A.; Javaid, M.; Singh, R.P.; Suman, R. Applications of Artificial Intelligence (AI) for cardiology during COVID-19 pandemic. *Sustain. Oper. Comput.* 2021, 2, 71–78. [CrossRef]
- 124. AL-Hashimi, M.; Hamdan, A. The Applications of Artificial Intelligence to Control COVID-19. In Advances in Data Science and Intelligent Data Communication Technologies for COVID-19; Springer: Cham, Switzerland, 2022; pp. 55–75.
- 125. Amaral, F.; Casaca, W.; Oishi, C.M.; Cuminato, J.A. Towards providing effective data-driven responses to predict the Covid-19 in São Paulo and Brazil. *Sensors* **2021**, *21*, 540. [CrossRef]
- 126. Zgheib, R.; Kamalov, F.; Chahbandarian, G.; El Labban, O. Diagnosing COVID-19 on Limited Data: A Comparative Study of Machine Learning Methods. In *International Conference on Intelligent Computing*; Springer: Cham, Switzerland, 2021; pp. 616–627.
- 127. Ferrari, L.; Gerardi, G.; Manzi, G.; Micheletti, A.; Nicolussi, F.; Biganzoli, E.; Salini, S. Modeling Provincial Covid-19 Epidemic Data Using an Adjusted Time-Dependent SIRD Model. *Int. J. Environ. Res. Public Health* **2021**, *18*, 6563. [CrossRef] [PubMed]

- 128. Almalki, Y.E.; Qayyum, A.; Irfan, M.; Haider, N.; Glowacz, A.; Alshehri, F.M.; Alduraibi, S.K.; Alshamrani, K.; Alkhalik Basha, M.A.; Alduraibi, A.; et al. A Novel Method for COVID-19 Diagnosis Using Artificial Intelligence in Chest X-ray Images. *Healthcare* 2021, 9, 522. [CrossRef] [PubMed]
- 129. Umair, M.; Khan, M.S.; Ahmed, F.; Baothman, F.; Alqahtani, F.; Alian, M.; Ahmad, J. Detection of COVID-19 Using Transfer Learning and Grad-CAM Visualization on Indigenously Collected X-ray Dataset. *Sensors* **2021**, *21*, 5813. [CrossRef] [PubMed]
- 130. Tamagusko, T.; Ferreira, A. Data-driven approach to understand the mobility patterns of the Portuguese population during the COVID-19 pandemic. *Sustainability* **2020**, *12*, 9775. [CrossRef]
- 131. Arvanitis, A.; Furxhi, I.; Tasioulis, T.; Karatzas, K. Prediction of the effective reproduction number of COVID-19 in Greece. A machine learning approach using Google mobility data. *medRxiv* 2021. [CrossRef]
- 132. Hussain, A.; Tahir, A.; Hussain, Z.; Sheikh, Z.; Gogate, M.; Dashtipour, K.; Ali, A.; Sheikh, A. Artificial intelligence-enabled analysis of public attitudes on facebook and twitter toward COVID-19 vaccines in the united kingdom and the united states: Observational study. *J. Med. Internet Res.* **2021**, *23*, e26627. [CrossRef]
- 133. Kumari, P.; Seeja, K.R. A novel periocular biometrics solution for authentication during Covid-19 pandemic situation. *J. Ambient. Intell. Humaniz. Comput.* **2021**, 12, 10321–10337. [CrossRef] [PubMed]
- 134. Talahua, J.S.; Buele, J.; Calvopiña, P.; Varela-Aldás, J. Facial Recognition System for People with and without Face Mask in Times of the COVID-19 Pandemic. *Sustainability* **2021**, *13*, 6900. [CrossRef]
- 135. Yu, X.; Lu, S.; Guo, L.; Wang, S.-H.; Zhang, Y.-D. ResGNet-C: A graph convolutional neural network for detection of COVID-19. *Neurocomputing* **2021**, *452*, 592–605. [CrossRef]
- Nayak, S.R.; Nayak, J.; Sinha, U.; Arora, V.; Ghosh, U.; Satapathy, S.C. An Automated Lightweight Deep Neural Network for Diagnosis of COVID-19 from Chest X-ray Images. *Arab. J. Sci. Eng.* 2021, 1–18. [CrossRef] [PubMed]
- 137. Bekhet, S.; Alkinani, M.H.; Tabares-Soto, R.; Hassaballah, M. An efficient method for COVID-19 detection using light weight convolutional neural network. *Comput. Mater. Contin.* **2021**, 69, 2475–2491. [CrossRef]
- 138. Keicher, M.; Burwinkel, H.; Bani-Harouni, D.; Paschali, M.; Czempiel, T.; Burian, E.; Makowski, M.R.; Braren, R.; Navab, N.; Wendler, T. U-GAT: Multimodal Graph Attention Network for COVID-19 Outcome Prediction. *arXiv* 2021, arXiv:2108.00860.
- 139. Alshazly, H.; Linse, C.; Barth, E.; Martinetz, T. Explainable COVID-19 detection using chest ct scans and deep learning. *Sensors* **2021**, *21*, 455. [CrossRef] [PubMed]
- 140. Carvalho, E.D.; Silva, R.; Araújo, F.; de Rabelo, R.; de Carvalho Filho, A.O. An approach to the classification of COVID-19 based on CT scans using convolutional features and genetic algorithms. *Comput. Biol. Med.* **2021**, *136*, 104744. [CrossRef]
- Fu, Y.; Xue, P.; Dong, E. Densely connected attention network for diagnosing COVID-19 based on chest CT. *Comput. Biol. Med.* 2021, 137, 104857. [CrossRef] [PubMed]
- Bougourzi, F.; Contino, R.; Distante, C.; Taleb-Ahmed, A. CNR-IEMN: A Deep Learning Based Approach to Recognise Covid-19 from CT-Scan. In Proceedings of the ICASSP 2021–2021 IEEE International Conference on Acoustics, Speech and Signal Processing (ICASSP), Toronto, ON, Canada, 6–11 June 2021; pp. 8568–8572.
- 143. Song, Y.; Zheng, S.; Li, L.; Zhang, X.; Zhang, X.; Huang, Z.; Chen, J.; Wang, R.; Zhao, H.; Chong, Y.; et al. Deep learning enables accurate diagnosis of novel coronavirus (COVID-19) with CT images. *IEEE/ACM Trans. Comput. Biol. Bioinform.* 2021, 18, 2775–2780. [PubMed]
- Alruwaili, M.; Shehab, A.; El-Ghany, A. COVID-19 Diagnosis Using an Enhanced Inception-ResNetV2 Deep Learning Model in CXR Images. J. Healthc. Eng. 2021, 2021, 6658058. [CrossRef] [PubMed]
- 145. Wang, S.; Kang, B.; Ma, J.; Zeng, X.; Xiao, M.; Guo, J.; Cai, M.; Yang, J.; Li, Y.; Meng, X.; et al. A deep learning algorithm using CT images to screen for Corona Virus Disease (COVID-19). *Eur. Radiol.* 2021, 31, 6096–6104. [CrossRef] [PubMed]
- 146. Jha, N.; Prashar, D.; Rashid, M.; Shafiq, M.; Khan, R.; Pruncu, C.I.; Saravana Kumar, M. Deep learning approach for discovery of in silico drugs for combating COVID-19. *J. Healthc. Eng.* **2021**, 2021, 6668985. [CrossRef]
- 147. Abbas, A.; Abdelsamea, M.M.; Gaber, M.M. Classification of COVID-19 in chest X-ray images using DeTraC deep convolutional neural network. *Appl. Intell.* 2021, *51*, 854–864. [CrossRef]
- 148. Sedik, A.; Hammad, M.; El-Samie, F.E.A.; Gupta, B.B.; El-Latif, A.A.A. Efficient deep learning approach for augmented detection of Coronavirus disease. *Neural Comput. Appl.* **2021**, 1–18. [CrossRef] [PubMed]
- 149. Bhardwaj, P.; Kaur, A. A novel and efficient deep learning approach for COVID-19 detection using X-ray imaging modality. *Int. J. Imaging Syst. Technol.* **2021**, 31, 1775–1791. [CrossRef]
- 150. Muneer, A.; Fati, S.M.; Akbar, N.A.; Agustriawan, D.; Wahyudi, S.T. iVaccine-Deep: Prediction of COVID-19 mRNA Vaccine Degradation using Deep Learning. *J. King Saud Univ.-Comput. Inf. Sci.* **2021**, *7*, e597. [CrossRef]
- 151. Ali, S.; Bello, B.; Patterson, M. Classifying COVID-19 Spike Sequences from Geographic Location Using Deep Learning. *arXiv* **2021**, arXiv:2110.00809.
- 152. Ahsan, M.; Based, M.; Haider, J.; Kowalski, M. COVID-19 detection from chest X-ray images using feature fusion and deep learning. *Sensors* **2021**, *21*, 1480.
- 153. Raji, P.; Prashantha, H.S.; Nisar, N.N.; Sulakhe, S.D.; Gita, S.; Shivani, M. Covid-19 Sleuthing using Pulmonary Radiography through CNN. In Proceedings of the 2021 Asian Conference on Innovation in Technology (ASIANCON), Pune, India, 27–29 August 2021; pp. 1–5.
- Teli, M.N. TeliNet: Classifying CT scan images for COVID-19 diagnosis. In Proceedings of the IEEE/CVF International Conference on Computer Vision, Montreal, BC, Canada, 11–17 October 2021; pp. 496–502.

- 155. Jacobs, S.A.; Moon, T.; McLoughlin, K.; Jones, D.; Hysom, D.; Ahn, D.H.; Gyllenhaal, J.; Watson, P.; Lightstone, F.C.; Allen, J.E.; et al. Enabling rapid COVID-19 small molecule drug design through scalable deep learning of generative models. *Int. J. High Perform. Comput. Appl.* 2021, 35, 469–482. [CrossRef]
- 156. Madhavan, M.V.; Khamparia, A.; Gupta, D.; Pande, S.; Tiwari, P.; Hossain, M.S. Res-CovNet: An internet of medical health things driven COVID-19 framework using transfer learning. *Neural Comput. Appl.* **2021**, 1–14. [CrossRef] [PubMed]
- 157. Shorfuzzaman, M. IoT-enabled stacked ensemble of deep neural networks for the diagnosis of COVID-19 using chest CT scans. *Computing* **2021**, 1–22. [CrossRef]
- 158. Shankar, K.; Perumal, E.; Díaz, V.G.; Tiwari, P.; Gupta; D.; Saudagar, A.K.J.; Muhammad, K. An optimal cascaded recurrent neural network for intelligent COVID-19 detection using Chest X-ray images. *Appl. Soft Comput.* **2021**, *113*, 107878. [CrossRef]
- 159. Sankaranarayanan, S.; Balan, J.; Walsh, J.R.; Wu; Y.; Minnich, S.; Piazza, A.; Osborne, C.; Oliver, G.R.; Lesko, J.; Bates, K.L.; et al. Covid-19 mortality prediction from deep learning in a large multistate electronic health record and laboratory information system data set: Algorithm development and validation. *J. Med. Internet Res.* **2021**, *23*, e30157. [CrossRef] [PubMed]
- Alhudhaif, A.; Polat, K.; Karaman, O. Determination of COVID-19 pneumonia based on generalized convolutional neural network model from chest X-ray images. *Expert Syst. Appl.* 2021, 180, 115141. [CrossRef]
- 161. Aboutalebi, H.; Pavlova, M.; Shafiee, M.J.; Sabri, A.; Alaref, A.; Wong, A. COVID-Net CXR-S: Deep Convolutional Neural Network for Severity Assessment of COVID-19 Cases from Chest X-ray Images. arXiv 2021, arXiv:2105.00256.
- Zhao, W.; Jiang, W.; Qiu, X. Fine-Tuning Convolutional Neural Networks for COVID-19 Detection from Chest X-ray Images. Diagnostics 2021, 11, 1887. [CrossRef]
- 163. Nguyen, D.C.; Ding, M.; Pathirana, P.N.; Seneviratne, A. Blockchain and AI-based solutions to combat coronavirus (COVID-19)like epidemics: A survey. *IEEE Access* 2021, *9*, 95730–95753.
- Fourati, L.C.; Samiha, A.Y.E.D. Federated Learning toward Data Preprocessing: COVID-19 Context. In Proceedings of the 2021 IEEE International Conference on Communications Workshops (ICC Workshops), Montreal, QC, Canada, 14–23 June 2021; pp. 1–6.
- 165. Baxter, G.; Beard, L.; Beattie, G.; Blake, M.; Greenhall, M.; Lingstadt, K.; Nixon, W.J.; Reimer, T. Covid-19 and the future of the digital shift amongst research libraries: An RLUK perspective in context. *New Rev. Acad. Librariansh.* 2021, 27, 322–348.
- 166. Houacine, N.A.; Drias, H. When robots contribute to eradicate the COVID-19 spread in a context of containment. *Prog. Artif. Intell.* **2021**, *10*, 391–416. [CrossRef]
- 167. Abd-alrazaq, A.A.; Alajlani, M.; Alhuwail, D.; Erbad, A.; Giannicchi, A.; Shah, Z.; Hamdi, M.; Househ, M. Blockchain technologies to mitigate COVID-19 challenges: A scoping review. *Comput. Methods Programs Biomed. Update* **2020**, *1*, 100001. [PubMed]
- 168. Zhang, C.; Xu, C.; Sharif, K.; Zhu, L. Privacy-preserving contact tracing in 5G-integrated and blockchain-based medical applications. *Comput. Stand. Interfaces* 2021, 77, 103520. [CrossRef] [PubMed]
- 169. Alabdulkarim, Y.; Alameer, A.; Almukaynizi, M.; Almaslukh, A. SPIN: A Blockchain-Based Framework for Sharing COVID-19 Pandemic Information across Nations. *Appl. Sci.* **2021**, *11*, 8767. [CrossRef]
- 170. Ghosh, B.; Biswas, A. Status evaluation of provinces affected by COVID-19: A qualitative assessment using fuzzy system. *Appl. Soft Comput.* **2021**, 109, 107540.
- 171. Deshpande, N.M.; Gite, S.S. A Brief Bibliometric Survey of Explainable AI in Medical Field. Libr. Philos. Pract. 2021, 1–27.
- 172. Javaid, M.; Khan, I.H. Internet of Things (IoT) enabled healthcare helps to take the challenges of COVID-19 Pandemic. *J. Oral Biol. Craniofacial Res.* 2021, *11*, 209–214. [CrossRef]
- 173. Liu, J.K.; Au, M.H.; Yuen, T.H.; Zuo, C.; Wang, J.; Sakzad, A.; Luo, X.; Li, L. Privacy-Preserving COVID-19 Contact Tracing App: A Zero-Knowledge Proof Approach. *IACR Cryptol. Eprint Arch.* **2020**, 2020, 528.
- 174. Serrano, J.C.M.; Papakyriakopoulos, O.; Hegelich, S. NLP-based feature extraction for the detection of COVID-19 misinformation videos on Youtube. In Proceedings of the 1st Workshop on NLP for COVID-19 at ACL, Online, July 2020.
- Lee, D.; Lee, J. Testing on the move: South Korea's rapid response to the COVID-19 pandemic. *Transp. Res. Interdiscip. Perspect.* 2020, 5, 100111. [CrossRef]
- Aggarwal, L.; Goswami, P.; Sachdeva, S. Multi-criterion intelligent decision support system for COVID-19. *Appl. Soft Comput.* 2021, 101, 107056. [CrossRef]
- 177. Huang, C.; Wang, M.; Rafaqat, W.; Shabbir, S.; Lian, L.; Zhang, J.; Lo, S.; Song, W. Data-driven Test Strategy for COVID-19 Using Machine Learning: A Study in Lahore, Pakistan. *Socio-Econ. Plan. Sci.* **2021**, 101091. [CrossRef] [PubMed]
- 178. Mohapatra, S.D.; Nayak, S.C.; Parida, S.; Panigrahi, C.R.; Pati, B. COVTrac: Covid-19 Tracker and Social Distancing App. In *Progress in Advanced Computing and Intelligent Engineering*; Springer: Singapore, 2021; pp. 607–619.
- 179. Ostropolets, A.; Zachariah, P.; Ryan, P.; Chen, R.; Hripcsak, G. Data Consult Service: Can we use observational data to address immediate clinical needs? *J. Am. Med. Informatics Assoc.* **2021**, *28*, 2139–2146. [CrossRef] [PubMed]
- Devi, M.; Maakar, S.K.; Sinwar, D.; Jangid, M.; Sangwan, P. Applications of Flying Ad-hoc Network During COVID-19 Pandemic. In Proceedings of the International Conference on Applied Scientific Computational Intelligence using Data Science (ASCI 2020), Jaipur, India, 22–23 December 2020; Volume 1099, p. 012005.
- Kerstens, K.; Shen, Z. Using COVID-19 mortality to select among hospital plant capacity models: An exploratory empirical application to Hubei province. *Technol. Forecast. Soc. Chang.* 2021, 166, 120535. [CrossRef] [PubMed]
- 182. Kumar, A.; Elsersy, M.; Darwsih, A.; Hassanien, A.E. Drones combat COVID-19 epidemic: Innovating and monitoring approach. In *Digital Transformation and Emerging Technologies for Fighting COVID-19 Pandemic: Innovative Approaches*; Springer: Cham, Switzerland, 2021; pp. 175–188.

- 183. Rouf, N.; Khan, A.K.; Malik, M.B.; Ud Din Khanday, A.M.; Gul, N. Unfolding the Potential of Impactful Emerging Technologies Amid COVID-19. In *Enabling Healthcare 4.0 for Pandemics: A Roadmap Using AI, Machine Learning, IoT and Cognitive Technologies*; Wiley: Hoboken, NJ, USA, 2021; p. 117.
- De Fazio, R.; Giannoccaro, N.I.; Carrasco, M.; Velazquez, R.; Visconti, P. Wearable devices and IoT applications for symptom detection, infection tracking, and diffusion containment of the COVID-19 pandemic: A survey. *Front. Inf. Technol. Electron. Eng.* 2021, 22, 1413–1442. [CrossRef]
- Mukati, N.; Namdev, N.; Dilip, R.; Hemalatha, N.; Dhiman, V.; Sahu, B. Healthcare assistance to COVID-19 patient using internet of things (IoT) enabled technologies. *Mater. Today Proc.* 2021. [CrossRef] [PubMed]
- 186. Lee, T.-Y.; Chen, Y.-L.; Fan, Y.-C. Next-Generation Electronics and Sensing Technology. Sensors 2021, 21, 7958. [CrossRef]
- 187. Pathak, N.; Deb, P.K.; Mukherjee, A.; Misra, S. IoT-to-the-Rescue: A Survey of IoT Solutions for COVID-19-like Pandemics. *IEEE Internet Things J.* 2021, *8*, 13145–13164. [CrossRef]
- Udgata, S.K.; Suryadevara, N.K. COVID-19, Sensors, and Internet of Medical Things (IoMT). In Internet of Things and Sensor Network for COVID-19; Springer: Singapore, 2021; pp. 39–53.
- Sharma, S.K.; Ahmed, S.S. IoT-based analysis for controlling & spreading prediction of COVID-19 in Saudi Arabia. *Soft Comput.* 2021, 25, 12551–12563.
- Khan, M.T.R.; Saad, M.M.; Tariq, M.A.; Akram, J.; Kim, D. SPICE-IT: Smart COVID-19 pandemic controlled eradication over NDN-IoT. *Inf. Fusion* 2021, 74, 50–64. [CrossRef]
- Awotunde, J.B.; Jimoh, R.G.; Matiluko, O.E.; Gbadamosi, B.; Ajamu, G.J. Artificial Intelligence and an Edge-IoMT-Based System for Combating COVID-19 Pandemic. In *Intelligent Interactive Multimedia Systems for e-Healthcare Applications*; Springer: Singapore, 2022; pp. 191–214.
- Abdulkareem, K.H.; Mohammed, M.A.; Salim, A.; Arif, M.; Geman, O.; Gupta, D.; Khanna, A. Realizing an effective COVID-19 diagnosis system based on machine learning and IOT in smart hospital environment. *IEEE Internet Things J.* 2021, *8*, 15919–15928. [CrossRef]
- 193. Jayachitra, V.P.; Nivetha, S.; Nivetha, R.; Harini, R. A cognitive IoT-based framework for effective diagnosis of COVID-19 using multimodal data. *Biomed. Signal Process. Control* 2021, 70, 102960. [CrossRef]
- Herath, H.M.K.K.M.B. Internet of Things (IoT) Enable Designs for Identify and Control the COVID-19 Pandemic. In Artificial Intelligence for COVID-19; Springer: Cham, Switzerland, 2021; pp. 423–436.
- 195. Mukherjee, R.; Kundu, A.; Mukherjee, I.; Gupta, D.; Tiwari, P.; Khanna, A.; Shorfuzzaman, M. IoT-cloud based healthcare model for COVID-19 detection: An enhanced k-Nearest Neighbour classifier based approach. *Computing* **2021**, 1–21. [CrossRef]
- 196. Akbarzadeh, O.; Baradaran, M.; Khosravi, M.R. IoT-based smart management of healthcare services in hospital buildings during COVID-19 and future pandemics. *Wirel. Commun. Mob. Comput.* **2021**, 2021, 5533161. [CrossRef]
- Petrović, N.; Kocić, Đ. IoT for COVID-19 Indoor Spread Prevention: Cough Detection, Air Quality Control and Contact Tracing. In Proceedings of the 2021 IEEE 32nd International Conference on Microelectronics (MIEL), Nis, Serbia, 12–14 September 2021; pp. 297–300.
- 198. Alamri, A.; Alamri, S. Live data analytics with IoT intelligence-sensing system in public transportation for COVID-19 pandemic. *Intell. Autom. Soft Comput.* 2021, 27, 441–452. [CrossRef]
- Poongodi, M.; Nguyen, T.N.; Hamdi, M.; Cengiz, K. A Measurement Approach Using Smart-IoT Based Architecture for Detecting the COVID-19. *Neural Process. Lett.* 2021, 1–15. [CrossRef] [PubMed]
- Kent, L.Y.; Kamsin, I.F.B. Implementation of IoT in Patient Health Monitoring and Healthcare for Hospitals. In Proceedings of the 3rd International Conference on Integrated Intelligent Computing Communication & Security (ICIIC 2021), Bangalore, India, 6–7 August 2021; pp. 473–479.
- Krishnan, R.S.; Kannan, A.; Manikandan, G.; Sri Sathya, K.B.; Sankar, V.K.; Narayanan, K.L. Secured College Bus Management System using IoT for Covid-19 Pandemic Situation. In Proceedings of the 2021 Third International Conference on Intelligent Communication Technologies and Virtual Mobile Networks (ICICV), Tirunelveli, India, 4–6 February 2021; pp. 376–382.
- Mylonas, G.; Amaxilatis, D.; Chatzigiannakis, I. Understanding the Effect of the COVID-19 Pandemic on the Usage of School Buildings in Greece Using an IoT Data-Driven Analysis. In Proceedings of the 2021 IEEE International Conference on Smart Internet of Things (SmartIoT), Jeju, Korea, 13–15 August 2021; pp. 365–370.
- Bhowmick, S.; Ferdous, T.; Momtaz, R. An IoT-Based Ambient Assisted Living for Elderly Care and Monitoring in COVID-19 Pandemic Using Arti Cial Intelligence and Deep Learning. Ph.D. Thesis, Brac University, Dhaka, Bangladesh, 2021.
- Herath, H.M.K.K.M.B.; Karunasena, G.M.K.B.; Herath, H.M.W.T. Development of an IoT Based Systems to Mitigate the Impact of COVID-19 Pandemic in Smart Cities. In *Machine Intelligence and Data Analytics for Sustainable Future Smart Cities*; Springer: Cham, Switzerland, 2021; pp. 287–309.
- 205. Herath, H.M.K.K.M.B.; Karunasena, G.M.K.B.; Madhusanka, B.G.D.A.; Priyankara, H.D.N.S. Internet of Medical Things (IoMT) Enabled TeleCOVID System for Diagnosis of COVID-19 Patients. In *Sustainability Measures for COVID-19 Pandemic*; Springer: Singapore, 2021; pp. 253–274.
- 206. Lastovicka-Medin, G.; Vanja, B. From Contactless Disinfection Intelligent Hand Sanitizer Dispenser for Public & Home towards IoT Based Assistive Technologies for Visually Impaired Users Institutional Responses to the COVID-19 Pandemic. In Proceedings of the 2021 10th Mediterranean Conference on Embedded Computing (MECO), Budva, Montenegro, 7–10 June 2021; pp. 1–4.
- 207. Rajasekar, S.J.S. An enhanced IoT based tracking model for COVID-19 cases. SN Comput. Sci. 2021, 2, 1–4. [CrossRef]

- 208. Alhmiedat, T.; Aborokbah, M. Social Distance Monitoring Approach Using Wearable Smart Tags. *Electronics* **2021**, *10*, 2435. [CrossRef]
- Puri, V.; Kataria, A.; Sharma, V. Artificial intelligence-powered decentralized framework for Internet of Things in Healthcare 4.0. *Trans. Emerg. Telecommun. Technol.* 2021, e4245.: 10.1002/ett.4245. [CrossRef]
- Alepis, E.; Maria, V.; Kontomaris, P. Covid-19 Mobile Tracking Application Utilizing Smart Sensors. In Proceedings of the 2021 12th International Conference on Information, Intelligence, Systems & Applications (IISA), Chania Crete, Greece, 12–14 July 2021; pp. 1–8.
- Ahuja, A.S.; Reddy, V.P.; Marques, O. Artificial intelligence and COVID-19: A multidisciplinary approach. *Integr. Med. Res.* 2020, 9, 100434. [CrossRef] [PubMed]
- 212. Márquez Díaz, J. Inteligencia artificial y Big Data como soluciones frente a la COVID-19. *Rev. BioéTica Derecho* 2020, *50*, 315–331. [CrossRef]
- 213. Anjum, N.; Asif, A.; Kiran, M.; Jabeen, F.; Yang, Z.; Huang, C.; Noor, S.; Imran, K.; Ali, I.; Mohamed, E.M. Intelligent COVID-19 forecasting, diagnoses and monitoring systems: A survey. *IEEE Commun. Surv. Tutorials* 2021, 14.
- Ahmad, A.; Bandara, M.; Fahmideh, M.; Proper, H.A.; Guizzardi, G.; Soar, J. An Overview of Ontologies and Tool Support for COVID-19 Analytics. In Proceedings of the 2021 IEEE 25th International Enterprise Distributed Object Computing Workshop (EDOCW), Gold Coast, Australia, 25–29 October 2021; pp. 1–8.
- Lainjo, B. The Enigmatic COVID-19 Vulnerabilities and the Invaluable Artificial Intelligence (AI). J. Multidiscip. Healthc. 2021, 14, 2361. [CrossRef] [PubMed]
- 216. Bazel, M.A.; Mohammed, F.; Alsabaiy, M.; Abualrejal, H.M. The role of Internet of Things, Blockchain, Artificial Intelligence, and Big Data Technologies in Healthcare to Prevent the Spread of the COVID-19. In Proceedings of the 2021 International Conference on Innovation and Intelligence for Informatics, Computing, and Technologies (3ICT), Zallaq, Bahrain, 29–30 September 2021; pp. 455–462.
- Swayamsiddha, S.; Prashant, K.; Shaw, D.; Mohanty, C. The prospective of Artificial Intelligence in COVID-19 Pandemic. *Health Technol.* 2021, 11, 1311–1320. [CrossRef] [PubMed]
- Deshpande, G.; Batliner, A.; Schuller, B.W. AI-Based human audio processing for COVID-19: A comprehensive overview. *Pattern Recognit.* 2021, 122, 108289. [CrossRef] [PubMed]
- 219. Naz, F.; Kumar, A.; Majumdar, A.; Agrawal, R. Is artificial intelligence an enabler of supply chain resiliency post COVID-19? An exploratory state-of-the-art review for future research. *Oper. Manag. Res.* **2021**, 1–21. [CrossRef]
- Pablo, R.G.-J.; Roberto, D.-P.; Victor, S.-U.; Isabel, G.-R.; Paul, C.; Elizabeth, O.-R. Big data in the healthcare system: A synergy with artificial intelligence and blockchain technology. J. Integr. Bioinform. 2021. [CrossRef]
- 221. Gupta, D.; Bhatt, S.; Gupta, M.; Tosun, A.S. Future smart connected communities to fight COVID-19 outbreak. *Internet Things* **2021**, *13*, 100342. [CrossRef]
- Maddikunta, P.K.R.; Pham, Q.-V.; Prabadevi, B.; Deepa, N.; Dev, K.; Gadekallu, T.R.; Ruby, R.; Liyanage, M. Industry 5.0: A survey on enabling technologies and potential applications. J. Ind. Inf. Integr. 2021, 100257. [CrossRef]
- Xu, X.; Lu, Y.; Vogel-Heuser, B.; Wang, L. Industry 4.0 and Industry 5.0—Inception, conception and perception. *J. Manuf. Syst.* 2021, 61, 530–535. [CrossRef]
- 224. Pappakrishnan, V.K.; Mythili, R.; Kavitha, V.; Parthiban, N. Role of Artificial Intelligence of Things (AIoT) in Covid-19 Pandemic: A Brief Survey. In Proceedings of the 6th International Conference on Internet of Things, Big Data and Security (IoTBDS 2021), Prague, Czech Republic, 23–25 April 2021.
- 225. Adadi, A.; Lahmer, M.; Nasiri, S. Artificial Intelligence and COVID-19: A Systematic Umbrella Review and Roads Ahead. J. King Saud-Univ.-Comput. Inf. Sci. 2021. [CrossRef]
- 226. Nadeem, O.; Saeed, M.S.; Tahir, M.A.; Mumtaz, R. A Survey of Artificial Intelligence and Internet of Things (IoT) based approaches against Covid-19. In Proceedings of the 2020 IEEE 17th International Conference on Smart Communities: Improving Quality of Life Using ICT, IoT and AI (HONET), Charlotte, NC, USA, 14–16 December 2020; pp. 214–218.
- 227. Abd-Alrazaq, A.; Alajlani, M.; Alhuwail, D.; Schneider, J.; Al-Kuwari, S.; Shah, Z.; Hamdi, M.; Househ, M. Artificial intelligence in the fight against COVID-19: Scoping review. *J. Med. Internet Res.* **2020**, *22*, e20756. [CrossRef] [PubMed]
- 228. Raza, K. Artificial intelligence against COVID-19: A meta-analysis of current research. In *Big Data Analytics and Artificial Intelligence Against COVID-19: Innovation Vision and Approach;* Springer: Cham, Switzerland, 2020; pp. 165–176.
- 229. Chen, J.; See, K.C. Artificial intelligence for COVID-19: Rapid review. J. Med. Internet Res. 2020, 22, e21476. [CrossRef] [PubMed]
- 230. Enughwure, A.A.; Febaide, I.C. Applications of artificial intelligence in combating Covid-19: A systematic review. *Open Access Libr. J.* **2020**, *7*, 1–12.
- Chiroma, H.; Ezugwu, A.E.; Jauro, F.; Al-Garadi, M.A.; Abdullahi, I.N.; Shuib, L. Early survey with bibliometric analysis on machine learning approaches in controlling coronavirus. *medRxiv* 2020. [CrossRef]
- Bullock, J.; Luccioni, A.; Pham, K.H.; Lam, C.S.N.; Luengo-Oroz, M. Mapping the landscape of artificial intelligence applications against COVID-19. J. Artif. Intell. Res. 2020, 69, 807–845.
- 233. Fong, S.J.; Dey, N.; Chaki, J. AI-enabled technologies that fight the coronavirus outbreak. In *Artificial Intelligence for Coronavirus Outbreak*; Springer: Singapore, 2021; pp. 23–45.
- Latif, S.; Usman, M.; Manzoor, S.; Iqbal, W.; Qadir, J.; Tyson, G.; Castro, I.; Razi, A.; Boulos, M.N.K.; Weller, A.; et al. Leveraging data science to combat covid-19: A comprehensive review. *IEEE Trans. Artif. Intell.* 2020, 1, 85–103. [CrossRef]

- 235. Chawki, M. Artificial Intelligence (AI) Joins the Fight Against COVID-19. In COVID-19: Prediction, Decision-Making, and Its Impacts; Springer: Singapore, 2021; pp. 1–7.
- Gunasekeran, D.V.; Tseng, R.M.W.W.; Tham, Y.-C.; Wong, T.Y. Applications of digital health for public health responses to COVID-19: A systematic scoping review of artificial intelligence, telehealth and related technologies. *NPJ Digit. Med.* 2021, 4, 1–6. [CrossRef] [PubMed]
- 237. Syeda, H.B.; Syed, M.; Sexton, K.W.; Syed, S.; Begum, S.; Syed, F.; Prior, F.; Yu, F., Jr. Role of machine learning techniques to tackle the COVID-19 crisis: Systematic review. *JMIR Med. Inform.* 2021, *9*, e23811. [CrossRef]
- 238. Zhao, Z.; Ma, Y.; Mushtaq, A.; Rajper, A.M.A.; Shehab, M.; Heybourne, A.; Song, W.; Ren, H.; Tse, Z.T.H. Applications of Robotics, AI, and Digital Technologies During COVID-19: A Review. *Disaster Med. Public Health Prep.* 2021, 1–23.
- 239. Kamalov, F.; Cherukuri, A.; Sulieman, H.; Thabtah, F.; Hossain, A. Machine learning applications for COVID-19: A state-of-the-art review. *arXiv* 2021, arXiv:2101.07824.
- 240. Safdari, R.; Rezayi, S.; Saeedi, S.; Tanhapour, M.; Gholamzadeh, M. Using data mining techniques to fight and control epidemics: A scoping review. *Health Technol.* **2021**, *7*, 1–13. [CrossRef]
- Nirmala, A.P.; More, S. Role of Artificial Intelligence in fighting against COVID-19. In Proceedings of the 2020 IEEE International Conference on Advances and Developments in Electrical and Electronics Engineering (ICADEE), Coimbatore, India, 10–11 December 2020; pp. 1–5.
- 242. Rasheed, J.; Jamil, A.; Hameed, A.A.; Al-Turjman, F.; Rasheed, A. COVID-19 in the Age of Artificial Intelligence: A Comprehensive Review. *Interdiscip. Sci. Comput. Life Sci.* 2021, 13, 1–23; 153–175. [CrossRef] [PubMed]
- Senthilraja, M. Application of artificial intelligence to address issues related to the COVID-19 Virus. Slas Technol. Transl. Life Sci. Innov. 2021, 26, 123–126. [CrossRef]
- 244. Kumar, V.; Singh, D.; Kaur, M.; Damaševičius, R. Overview of current state of research on the application of artificial intelligence techniques for COVID-19. *PeerJ Comput. Sci.* 2021, 7, e564. [CrossRef]
- Chen, J.; Li, K.; Zhang, Z.; Li, K.; Yu, P.S. A survey on applications of artificial intelligence in fighting against COVID-19. ACM Comput. Surv. (CSUR) 2021, 54, 1–32. [CrossRef]
- Dogan, O.; Tiwari, S.; Jabbar, M.A.; Guggari, S. A systematic review on AI/ML approaches against COVID-19 outbreak. *Complex Intell. Syst.* 2021, 7, 2655–2678. [CrossRef] [PubMed]
- 247. Alafif, T.; Tehame, A.M.; Bajaba, S.; Barnawi, A.; Zia, S. Machine and deep learning towards COVID-19 diagnosis and treatment: Survey, challenges, and future directions. *Int. J. Environ. Res. Public Health* **2021**, *18*, 1117. [CrossRef] [PubMed]
- 248. Singh, B.; Datta, B.; Ashish, A.; Dutta, G. A comprehensive review on current COVID-19 detection methods: From lab care to point of care diagnosis. *Sensors Int.* 2021, 2, 100119. [CrossRef]
- 249. Tang, B.; He, F.; Liu, D.; Fang, M.; Wu, Z.; Xu, D. AI-aided design of novel targeted covalent inhibitors against SARS-CoV-2. *BioRxiv* 2020. [CrossRef]
- Zhavoronkov, A.; Aladinskiy, V.; Zhebrak, A. Potential COVID-2019 3C-like protease inhibitors designed using generative deep learning approaches. *Insilico Med.* 2020, 307, E1.
- Hofmarcher, M.; Mayr, A.; Rumetshofer, E.; Ruch, P. Large-Scale Ligand-Based Virtual Screening for SARSCoV-2 Inhibitors Using Deep Neural Networks. Soc. Sci. Res. Netw. 2020.
- Majeed, A. Effective Handling of COVID-19 Pandemic: Experiences and Lessons from the Perspective of South Korea. COVID 2021, 1, 325–334. [CrossRef]
- Majeed, A.; Hwang, S.O. A Comprehensive Analysis of Privacy Protection Techniques Developed for COVID-19 Pandemic. *IEEE Access* 2021, 9, 164159-164187. [CrossRef]
- Blasch, E.; Pham, T.; Chong, C.-Y.; Koch, W.; Leung, H.; Braines, D.; Abdelzaher, T. Machine Learning/Artificial Intelligence for Sensor Data Fusion–Opportunities and Challenges. *IEEE Aerosp. Electron. Syst. Mag.* 2021, 36, 80–93. [CrossRef]
- 255. Okereafor, K. Cybersecurity in the COVID-19 Pandemic; CRC Press: Boca Raton, FL, USA, 2021.
- 256. Mondaini, L.; Meirose, B.; Mondaini, F. Second quantization approach to COVID-19 epidemic. arXiv 2021, arXiv:2104.04602.
- Awotunde, J.B.; Jimoh, R.G.; Oladipo, I.D.; Abdulraheem, M.; Jimoh, T.B.; Ajamu, G.J. Big Data and Data Analytics for an Enhanced COVID-19 Epidemic Management. In *Artificial Intelligence for COVID-19*; Springer: Cham, Switzerland, 2021; pp. 11–29.
   Control A. Nill J. P. Drive Analytics in Mathematical Intelligence for COVID-19; Springer: Cham, Switzerland, 2021; pp. 11–29.
- 258. Gupta, A.; Nihal, P. Data Analytics and Artificial Intelligence—A Boon for Start-Ups. Emerge 2020, 21, 35.
- Cortés, U.; Cortés, A.; Garcia-Gasulla, D.; Pérez-Arnal, R.; Álvarez-Napagao, S.; Álvarez, E. The ethical use of high-performance computing and artificial intelligence: Fighting COVID-19 at Barcelona Supercomputing Center. AI Ethics 2021, 1–16. [CrossRef] [PubMed]
- 260. Kiliç, M. Ethico-Juridical Dimension of Artificial Intelligence Application in the Combat to Covid-19 Pandemics. In *The Impact of Artificial Intelligence on Governance, Economics and Finance*; Springer: Singapore, 2021; Volume 1, pp. 299–317.
- Delanerolle, G.; Rathod, S.; Elliot, K.; Ramakrishnan, R.; Thayanandan, T.; Sandle, N.; Haque, N.; Raymont, V.; Phiri, P. Rapid commentary: Ethical implications for clinical trialists and patients associated with COVID-19 research. *World J. Psychiatry* 2021, 11, 58. [CrossRef] [PubMed]
- Awotunde, J.B.; Ogundokun, R.O.; Misra, S. Cloud and IoMT-based big data analytics system during COVID-19 pandemic. In Efficient Data Handling for Massive Internet of Medical Things; Springer: Cham, Switzerland, 2021; pp. 181–201.
- Ghimire, A.; Thapa, S.; Jha, A.K.; Kumar, A.; Kumar, A.; Adhikari, S. AI and IoT solutions for tackling COVID-19 pandemic. In Proceedings of the 2020 4th International Conference on Electronics, Communication and Aerospace Technology (ICECA), 5–7 November 2020, Coimbatore, India., pp. 1083–1092.

- Siriwardhana, Y.; De Alwis, C.; Gür, G.; Ylianttila, M.; Liyanage, M. The fight against the COVID-19 pandemic with 5G technologies. *IEEE Eng. Manag. Rev.* 2020, 48, 72–84. [CrossRef]
- Elbasi, E.; Topcu, A.E.; Mathew, S. Prediction of COVID-19 Risk in Public Areas Using IoT and Machine Learning. *Electronics* 2021, 10, 1677. [CrossRef]
- 266. Shin, D. The effects of explainability and causability on perception, trust, and acceptance: Implications for explainable AI. Int. J. Hum. Comput. Stud. 2021, 146, 102551. [CrossRef]
- 267. Kundu, S. AI in medicine must be explainable. Nat. Med. 2021, 27, 1328. [CrossRef]
- Meske, C.; Bunde, E.; Schneider, J.; Gersch, M. Explainable artificial intelligence: Objectives, stakeholders, and future research opportunities. *Inf. Syst. Manag.* 2021, 39, 1–11. [CrossRef]
- Ehsan, U.; Liao, Q.V.; Muller, M.; Riedl, M.O.; Weisz, J.D. Expanding explainability: Towards social transparency in ai systems. In Proceedings of the 2021 CHI Conference on Human Factors in Computing Systems, Yokohama, Japan, 8–13 May 2021; pp. 1–19.
- 270. Ye, Q.; Xia, J.; Yang, G. Explainable AI For COVID-19 CT Classifiers: An Initial Comparison Study. arXiv 2021, arXiv:2104.14506.
- Bakheet, S.; Al-Hamadi, A. Automatic detection of COVID-19 using pruned GLCM-Based texture features and LDCRF classification. *Comput. Biol. Med.* 2021, 137, 104781. [CrossRef] [PubMed]
- 272. Coelho, Y.; Lampier, L.; Valadão, C.; Caldeira, E.; Delisle-Rodríguez, D.; Villa-Parra, A.C.; Cobos-Maldonado, C.; Calle-Siguencia, J.; Urgiles-Ortiz, F.; Bastos-Filho, T. Towards the Use of Artificial Intelligence Techniques in Biomedical Data from an Integrated Portable Medical Assistant to Infer Asymptomatic Cases of COVID-19. In *International Conference on Information Technology & Systems*; Springer: Cham, Switzerland, 2021.
- Sudir, P.; Hanumantharaju, M.C.; Aradhya, V.N.M. Efficient COVID-19 Diagnosis Approach Using Multi-scale Retinex and Convolution Neural Network. In *Data Engineering and Intelligent Computing*; Springer: Singapore, 2021; pp. 523–530.
- 274. Connor, S.; Khoshgoftaar, T.M.; Borko, F. Deep Learning applications for COVID-19. J. Big Data 2021, 8, 1–54.
- Basiri, M.E.; Nemati, S.; Abdar, M.; Asadi, S.; Acharrya, U.R. A novel fusion-based deep learning model for sentiment analysis of COVID-19 tweets. *Knowl.-Based Syst.* 2021, 228, 107242. [CrossRef]
- 276. Lopez, C.E.; Gallemore, C. An augmented multilingual Twitter dataset for studying the COVID-19 infodemic. Soc. Netw. Anal. Min. 2021, 11, 1–14. [CrossRef]
- 277. Luo, Y.; Xu, X. Comparative study of deep learning models for analyzing online restaurant reviews in the era of the COVID-19 pandemic. *Int. J. Hosp. Manag.* 2021, 94, 102849. [CrossRef]
- 278. Ghasiya, P.; Okamura, K. Investigating COVID-19 News Across Four Nations: A Topic Modeling and Sentiment Analysis Approach. *IEEE Access* 2021, 9, 36645–36656. [CrossRef]
- Wang, P.; Shi, H.; Wu, X.; Jiao, L. Sentiment Analysis of Rumor Spread Amid COVID-19: Based on Weibo Text. *Healthcare* 2021, 9, 1275. [CrossRef] [PubMed]
- 280. Alorini, G.; Rawat, D.B.; Alorini, D. LSTM-RNN Based Sentiment Analysis to Monitor COVID-19 Opinions using Social Media Data. In Proceedings of the ICC 2021-IEEE International Conference on Communications, 14–23 June 2021, Montreal, Canada.
- 281. Aljameel, S.S.; Alabbad, D.A.; Alzahrani, N.A.; Alqarni, S.M.; Alamoudi, F.A.; Babili, L.M.; Aljaafary, S.K.; Alshamrani, F.M. A sentiment analysis approach to predict an individual's awareness of the precautionary procedures to prevent COVID-19 outbreaks in Saudi Arabia. *Int. J. Environ. Res. Public Health* 2021, 18, 218. [CrossRef] [PubMed]
- Xie, X.; Siau, K.; Nah, F.F.-H. COVID-19 pandemic—Online education in the new normal and the next normal. J. Inf. Technol. Case Appl. Res. 2020, 22, 175–187. [CrossRef]
- Tilli, M.; Olliaro, P.; Gobbi, F.; Bisoffi, Z.; Bartoloni, A.; Zammarchi, L. Neglected tropical diseases in non-endemic countries in the era of COVID-19 pandemic: The great forgotten. *J. Travel Med.* 2021, 28, taaa179. [CrossRef] [PubMed]
- Lee, S.M.; Lee, D. Opportunities and challenges for contactless healthcare services in the post-COVID-19 Era. *Technol. Forecast.* Soc. Chang. 2021, 167, 120712. [CrossRef] [PubMed]
- Iqbal, M.Z.; Campbell, A.G. From luxury to necessity: Progress of touchless interaction technology. *Technol. Soc.* 2021, 67, 101796. [CrossRef]
- Wibowo, D.; Fahmi, F. Contactless and Cashless Smart Vending Machine Integrated with Mobile Device. In Proceedings of the 2021 5th International Conference on Electrical, Telecommunication and Computer Engineering (ELTICOM), Sumatra, Indonesia, 15–16 September, 2021; Volume 5, pp. 146–151.
- Ajmal, M.M.; Khan, M.; Shad, M.K.; AlKatheeri, H.; Jabeen, F. Socio-economic and technological new normal in supply chain management: Lessons from COVID-19 pandemic. *Int. J. Logist. Manag.* 2021. [CrossRef]
- Valdes-Sosa, P.A.; Evans, A.C.; Valdes-Sosa, M.J.; Poo, M. A call for international research on COVID-19-induced brain dysfunctions. *Natl. Sci. Rev.* 2021. [CrossRef]
- 289. Sarría-Santamera, A.; Yeskendir, A.; Maulenkul, T.; Orazumbekova, B.; Gaipov, A.; Imaz-Iglesia, I.; Pinilla-Navas, L.; Moreno-Casbas, T.; Corral, T. Population health and health services: Old challenges and new realities in the COVID-19 era. *Int. J. Environ. Res. Public Health* 2021, 18, 1658. [CrossRef]
- 290. Alqudaihi, K.S.; Aslam, N.; Khan, I.U.; Almuhaideb, A.M.; Alsunaidi, S.J.; Ibrahim, N.M.A.R.; Alhaidari, F.A.; Shaikh, F.S.; Alsenbel, Y.M.; Alalharith, D.M.; et al. Cough Sound Detection and Diagnosis Using Artificial Intelligence Techniques: Challenges and Opportunities. *IEEE Access* 2021, 9, 102327–102344. [CrossRef] [PubMed]
- Alhasan, M.; Hasaneen, M. Digital Imaging, Technologies and Artificial Intelligence Applications during COVID-19 pandemic. Comput. Med. Imaging Graph. 2021, 91, 101933. [CrossRef]

- 292. Sun, C.-C. Analyzing Determinants for Adoption of Intelligent Personal Assistant: An Empirical Study. *Appl. Sci.* 2021, *11*, 10618. [CrossRef]
- 293. Papachristou, S.; Stamatiou, I.; Stoian, A.P.; Papanas, N. New-onset diabetes in COVID-19: Time to frame its fearful symmetry. *Diabetes Ther.* 2021, 12, 461–464. [CrossRef] [PubMed]
- Pandey, S.R.; Nguyen, M.N.H.; Dang, T.N.; Tran, N.H.; Thar, K.; Han, Z.; Hong, C.S. Edge-assisted Democratized Learning Towards Federated Analytics. *IEEE Internet Things J.*, 2021. [CrossRef]
- 295. Wang, Z.; Zhu, Y.; Wang, D.; Han, Z. FedACS: Federated Skewness Analytics in Heterogeneous Decentralized Data Environments. In Proceedings of the 2021 IEEE/ACM 29th International Symposium on Quality of Service (IWQOS), Tokyo, Japan 25–28 June 2021; pp. 1–10.
- 296. Zhang, W.; Zhou, T.; Lu, Q.; Wang, X.; Zhu, C.; Sun, H.; Wang, Z.; Lo S.K.; Wang, F.Y. Dynamic fusion-based federated learning for COVID-19 detection. *IEEE Internet of Things J.* 2021, *8*, 15884–15891. [CrossRef]
- 297. Nguyen, D.C.; Ding, M.; Pathirana, P.N.; Seneviratne, A.; Zomaya, A.Y. Federated Learning for COVID-19 Detection with Generative Adversarial Networks in Edge Cloud Computing. *IEEE Internet Things J.* **2021**. [CrossRef]
- Pang, J.; Huang, Y.; Xie, Z.; Li, J.; Cai, Z. Collaborative city digital twin for the COVID-19 pandemic: A federated learning solution. *Tsinghua Sci. Technol.* 2021, 26, 759–771. [CrossRef]
- 299. Dayan, I.; Roth, H.R.; Zhong, A.; Harouni, A.; Gentili, A.; Abidin, A.Z.; Liu, A.; Costa, A.B.; Wood, B.J.; Tsai, C.S.; et al. Federated learning for predicting clinical outcomes in patients with COVID-19. *Nat. Med.* **2021**, *27*, 1735–1743. [CrossRef] [PubMed]